A Review of the Use of Geostationary Satellite Observations in Regional-Scale Models for Short-term Cloud Forecasting

FREDERIK KURZROCK^{1,2*}, SYLVAIN CROS², FABRICE CHANE MING³, JASON A. OTKIN⁴, AXEL HUTT^{5,6}, LAURENT LINGUET⁷, GILLES LAJOIE¹ and ROLAND POTTHAST^{5,6}

¹Institut de Recherche pour le Développement (IRD), UMR 228 ESPACE-DEV, Université de La Réunion, Sainte Clotilde, La Réunion, France

²Reuniwatt SAS, Sainte Clotilde, La Réunion, France

³Laboratoire de l'Atmosphère et des Cyclones, UMR8105, UMR CNRS – Météo-France – Université, Université de La Réunion, La Réunion, France

⁴Space Science and Engineering Center, Cooperative Institute for Meteorological Satellite Studies, University of Wisconsin-Madison, Madison, WI, USA

⁵Deutscher Wetterdienst, Offenbach, Germany

⁶Department of Mathematics, University of Reading, Reading, United Kingdom

⁷Institut de Recherche pour le Développement (IRD), UMR 228 ESPACE-DEV, Université de Guyane, Cayenne, Guyane, France

(Manuscript received January 15, 2018; in revised form May 17, 2018; accepted May 22, 2018)

Abstract

Many research and societal applications such as surface solar irradiance assessment and forecasting require accurate short-term cloudiness forecasts at kilometre and hourly scales. Today limited-area numerical weather prediction models have the potential to provide such forecasts by simulating clouds at high spatial and temporal resolutions. However, the forecast performance during the first 12-24 h is strongly influenced by the accuracy of the cloud and thermodynamic analyses in the initial conditions. Geostationary meteorological satellites provide valuable observations that can be used in data assimilation for frequent cloud analysis determination. This paper provides an up-to-date review of the state of the art in cloud-related geostationary satellite data assimilation with limited-area models dedicated to improve cloudiness forecast performance. Research and operational studies have been reviewed by differentiating between satellite radiance and cloud property retrieval assimilation. This review gives insight into the best practices considering the large variety of limited-area models, data assimilation methods, satellite sensors and channels, cloud property retrieval products and various methodological challenges. Cloud analysis methods for regional models have become more sophisticated in recent years and are increasingly able to exploit observations from geostationary satellites. Important proofs of concept have been performed in this decade, paving the way for an optimal synergy of geostationary satellite data assimilation and convection-permitting limited-area model forecasts. At the same time, the increasing amount of channels of geostationary satellite instruments leads to more opportunities and challenges for data assimilation methods.

Keywords: satellite data assimilation, geostationary satellites, limited-area models, cloudiness forecasting

1 Introduction

Predicting the occurrence and evolution of cloud systems is a key component of atmospheric science and numerical weather prediction (NWP). Moreover, cloudiness forecasts are used in research, defence and societal applications.

Research-wise, continuous efforts are being made to improve the representation of clouds in global and regional NWP models used for weather forecasting and climate projections. Cloudiness forecasts greatly influence other model components such as land surface or hydrological models (FORMAN and MARGULIS, 2010; MEETSCHEN et al., 2004). Numerous defence applications require accurate sky conditions on the operation terrain, especially the presence of a cloud free line of sight. Remote monitoring systems using optical and thermal vision, optical communication between ground and airplane or satellite and optronic aiming sight are not operational in cloudy conditions (CROS et al., 2015). Finally, cloud cover presence must also be forecasted for societal applications. Transportation security (e.g. air traffic and roads) needs accurate information about cloud base height, fog, icing and precipitation. Within the health system, cloud information is needed to assess air pollution and ultraviolet exposure.

A wide range of economic sectors are also influenced by cloud forecasts. For example, some leisure activities, such as sailing, trekking, paragliding, photography

^{*}Corresponding author: Frederik Kurzrock Institut de Recherche pour le Développement (IRD), UMR 228 ESPACE-DEV, Université de La Réunion, 15 avenue René Cassin, 97715 ST DENIS Cedex 9, La Réunion, France e-Mail: frederik.kurzrock@univ-reunion.fr

or outdoor shooting, are strongly dependent on cloud cover. More and more tourism and leisure industries are taking such forecasts into account in their current operations and are able to quantify the added economic value of improved cloudiness forecasts.

A currently growing application is solar power forecasting, which requires very accurate and frequently updated cloudiness forecasts. Photovoltaic (PV) power production variations are mainly driven by the attenuation of incoming solar radiation due to clouds passing between the sun and the PV panels. The simulation of cloud optical properties over minute to hourly time scales and at a spatial scale of a few kilometres is necessary to accurately forecast irradiance reaching the Earth's surface (SENGUPTA et al., 2015). Therefore, PV production management is particularly demanding in terms of cloud forecast performance and can be considered as a guiding application to improve it. For this application, NWP is currently the most suitable solution to produce cloud forecasts for lead times beyond 6 h up to several days ahead (DIAGNE et al., 2013). Due to an incomplete representation of the complex and often non-linear cloud processes and associated impacts on radiation, NWP models tend to underestimate lowlevel cloud cover and thus overpredict solar irradiance at the surface. This concerns especially stratus clouds and coastal areas (HAIDEN and TRENTMANN, 2015; INMAN et al., 2013; YANG and KLEISSL, 2016; YUCEL et al., 2002). There are also large model uncertainties regarding the evolution of upper level cloudiness (CINTINEO et al., 2014).

Nevertheless, the temporal and spatial resolutions of global model output (typically more than 10 km and 1 hour) often do not permit accurate cloud forecasting since cloud processes occur over shorter temporal and spatial scales. The parameterisation of sub-grid scale processes induces important uncertainties of irradiance forecasts since these processes often cause a highly variable cloud cover within a given grid box. SENGUPTA et al. (2015) gives a comprehensive overview of global models and their utility for solar irradiance forecasting. BENGTSSON et al. (2013) evaluate and discuss the benefit of introducing stochastic elements into convective parameterizations.

Among current initiatives to improve NWP models for solar power forecasting, the WRF-Solar project aims at taking the interactions between clouds, aerosols and radiation in the regional NWP model (or limitedarea model (LAM)) WRF (Weather Research and Forecasting) better into account (JIMENEZ et al., 2016). Besides, the choice and combination of parameterisation schemes (e.g. convection, cloud microphysics, radiation and planetary boundary layer) also highly influences the performance of NWP models (CINTINEO et al., 2014; LÓPEZ-COTO et al., 2013; OTKIN et al., 2017; OTKIN and GREENWALD, 2008).

An area of research with great potential for improving cloud forecasts produced by NWP models is data assimilation (DA) (GEER et al., 2017). DA is used to determine the most likely state of the atmosphere at a given time, with the resultant analysis providing the initial conditions for a NWP model forecast. The atmospheric initial state is determined by an optimal combination of background information (usually short-range NWP forecasts which are also called first guess) and atmospheric observations (KALNAY, 2003).

Satellites are the primary source of cloud information, and therefore play an important role for cloud analysis determination. By virtue of their high spatial (< 4 km) and temporal (5–15 min) resolutions, geostationary satellite observations are an ideal data source for many of the above mentioned applications of regional cloudiness forecasting, such as weather nowcasting, military applications (RUGGIERO et al., 1999) and solar power forecasting (CROS et al., 2014; LORENZ et al., 2014).

However, DA constitutes a wide and complex research domain where various efforts around the globe have been made in the past in order to improve different specific aspects of weather forecasting. BAUER et al. (2011b) list recommendations for improving various aspects regarding the assimilation of satellite-based cloud observations, such as modelling, verification, data assimilation and the exploitation of observations, and call for more collaboration between the different communities. Despite recent progress in the field, the potential for cloud data assimilation in cloud resolving NWP models using geostationary sensors is far from being fully exploited (GUSTAFSSON et al., 2018). Many issues faced within limited-area models are different than those faced in global models that rely far more heavily on polarorbiting satellite sensors.

Consequently, a detailed review of geostationary satellite data assimilation in regional-scale NWP models for cloudiness forecasting will provide guidance for further research. In this paper, we will identify and describe the most promising methods and their associated limitations, paving the way for developments aimed at improving cloudiness forecasts for demanding applications such as solar energy management.

Section 2 introduces the concepts of cloudiness forecasting using regional NWP with geostationary data assimilation approaches. While Section 3 details the use of radiances as assimilated observations, Section 4 presents the recent efforts concerning the assimilation of satellite-based cloud properties. Section 5 discusses different comprehensive aspects of current research and operational cloud analysis systems. Conclusions are drawn in Section 6.

2 Satellite data assimilation for cloudiness forecasts: basic concepts

LAMs obtain their lateral and initial boundary conditions from global circulation models (GCMs). This allows LAMs to use higher temporal and spatial resolutions than GCMs and potentially simulate cloud processes with more detail. Observations can be taken into



Figure 1: Two-dimensional (horizontal) sketch of different representations of a partly cloudy region as seen from above. The grey areas represent cloud presence while white means no cloud. a) "True" situation of cloud presence. b) Cloud presence observation as seen by the satellite. c) Example of a first guess produced by a regional NWP model. d) Mapping of the satellite observation to the model grid. e) Cloud analysis resulting from c) and d) using data assimilation.

account to produce a more realistic initial representation of the three-dimensional atmospheric state on the whole grid of the LAM. But even with a dense observation network, observations would not be available for every grid box, therefore necessitating the use of DA methods that can spread observational information to other parts of the domain and to unobserved model variables. DA offers the possibility to determine the most realistic estimates of the atmospheric state in the domain of the LAM (the so-called "analysis") by combining the unevenly distributed atmospheric and land surface observations with previous model estimates (the so-called "background" or "first guess").

Determining an atmospheric analysis in the presence of clouds is a complex issue because clouds are captured differently by observations and NWP models. Fig. 1 can be understood as a two-dimensional illustrative example of this issue. The goal is to build a two-dimensional cloud analysis (e) with a certain grid spacing containing the simplified information whether clouds are present or not in a given grid box. In this example the "truth" (a) is a situation with four cloud systems. It is observed by a satellite (b) with a certain resolution and error. A NWP model provides a first guess of the situation (c) with the respective model grid spacing. The satellite observations are mapped to the model grid (d). In the actual DA procedure the cloud analysis (e) is determined by taking into account the first guess, the mapped observations and their respective uncertainties. In reality the situation is much more complex since the vertical and temporal dimensions and other cloud-related parameters like cloud fraction, phase, optical thickness or precipitation are also important and may be part of the cloud analysis that is to be determined.

In the case of cloudiness forecasting, a precise cloud analysis is a priority compared to non-related parameters (e.g. ozone concentration or land surface parameters). Thus, depending on the application of the regional modelling efforts, it is appropriate to focus on the assimilation of a subset of observation types that are directly sensitive to clouds or water vapour. For the forecasting of clouds, these are most commonly satellite radiances or retrieved cloud properties. Global DA experiments have shown that the skill of weather forecasts largely depends on the accuracy of the initial conditions in cloud-covered areas (MCNALLY, 2002). PINCUS et al. (2011) performed perfect model experiments with two global models that showed that the assimilation of cloud information is especially advantageous in regions where other observations are sparse.

In the past decades, a variety of data assimilation methods and strategies, such as nudging, optimal interpolation, variational methods such as 3D-Var, 4D-Var, and ensemble Kalman filters (EnKF) have been developed and successfully applied to numerous global and regional NWP models by diverse research institutions and weather offices around the world (GUSTAFSSON et al., 2018). In variational methods the minimisation of a cost function has to be found in order to determine the analysis. As an example, the cost function of the 3D-Var method is given by:

$$J(x) = (x - x_b)^T B^{-1} (x - x_b) + (y - H(x))^T R^{-1} (y - H(x))$$
(2.1)

With x being the analysis, x_b the background, y the observations, B the background error covariance matrix, R the observation error covariance matrix and H the observation operator (or forward operator) which maps the model state to the observation space. A radiative transfer model (RTM) that converts model fields (i.e. meteorological variables) into synthetic satellite radiances is a common example of an observation operator. The analysis can be determined by using available observations and a background in a cycling procedure.

Two of the most-widely used DA methods are 4D-Var and EnKF. While 3D-Var considers the three spatial dimensions and aims at minimising the cost function for a given analysis time, the 4D-Var method also includes the temporal dimension. The evolution of the NWP model and the observations is considered during a certain timeframe before the analysis time (usually several hours) in order to derive the atmospheric analysis. This timeframe is called the "assimilation window". The assimilation window length has to be restricted to a few hours since 4D-Var assumes linear dynamic processes because of the need for an adjoint model and also that the NWP model is perfect. These issues limit the utility of this method for high-resolution simulations and therefore motivate using the EnKF. The use of an ensemble allows for better quantification of flow-dependent covariances important for clouds and also allows fully nonlinear forward observation operators. Ensemble forecasting provides a sample of atmospheric states which can be used to estimate the evolution of the mean state and the covariance or the model background uncertainty.

More detailed explanations about DA methods in general are given by KALNAY (2003). The advantages, disadvantages and differences between variational, ensemble-based methods and hybrid methods that combine aspects of variational and ensemble DA are discussed in BANNISTER (2017) and KALNAY et al. (2007). A detailed review of the EnKF is given by HOUTEKAMER and ZHANG (2016). We also mention that recent particle filter developments may improve the data assimilation (ZHU et al., 2017).

Geostationary meteorological satellites provide atmospheric observations on a global scale that are highly valuable for the determination of cloud analyses. Indeed, this source of observation is the only one offering pertinent information about cloud presence, properties and evolution with high spatio-temporal resolution and large geographic coverage. Thus, the assimilation of geostationary satellite observations in regional NWP models satisfies the demanding requirements of some applications such as solar power forecasting. It is worth mentioning that the resolution of geostationary satellite observations decreases with increasing distance from the subsatellite point over the Equator. This circumstance is especially important with increasing latitude (e.g. Scandinavian countries) since in the tropics it can simply be overcome by choosing an appropriate satellite.

The on-board sensors of these satellites measure radiances at different wavelengths or channels, i.e. in the infrared (IR), near infrared (NIR) and visible (VIS) spectrum. The IR channels are often distinguished between those that are primarily sensitive either to water vapour (WV), temperature, atmospheric trace gases (e.g. ozone) or window channels. Window channels provide useful information about cloud top properties when clouds are present and about the surface when clouds are missing from a given scene. Combining spectral signatures from multiple IR, VIS and NIR channels allows for the estimation of many cloud properties, such as cloud top height, phase and optical depth.

Depending on their spectral characteristics, the assimilation of different channels involves different problems. For example, the successful assimilation of window channels requires accurate information about the land-surface emissivity, which is difficult to obtain. Because of this, such observations are typically excluded in satellite DA (HARNISCH et al., 2016). Most observations obtained from geostationary satellites for a given location are fundamentally different depending on the presence of clouds (Fig. 2). The figure depicts that different channels are sensitive to different levels of the atmosphere. The vertical height of their maximum sensitivity strongly depends on the presence and vertical location of clouds. While the yellow sensitivity curve in Fig. 2 serves as an example of a water vapour channel, the red and blue curve are examples of the sensitivity of



Figure 2: Sketches of the sensitivity of two different satellite channels in the clear-sky and cloudy case. Yellow: channel A is generally unaffected by the cloud; Red: channel B is affected by cloud; Blue: channel B in the clear sky case.

a thermal IR channel in the cloud-free and cloudy case. Either information about the temperature of the lower atmosphere or of the cloud top can be derived. The sensitivity curve of a VIS channel would either peak at the surface or at the cloud top.

Satellites do not measure quantities like temperature or humidity, but radiances, which are radiometric signals. These can either be assimilated directly or converted into physical cloud properties, called "retrievals", before their assimilation. In radiance assimilation the NWP model variables must be converted into synthetic radiances while in retrieval assimilation the satellitebased variables are converted into model variables. Both techniques have the potential to significantly increase the cloud-related information content of an analysis and thus contribute to better short-term forecasts of cloud features.

The following two sections provide a review of recent efforts in the two fields of radiance and cloud property assimilation. Current methods are assessed regarding their potential to improve the accuracy of highresolution limited-area short-term forecasts of cloudrelated parameters like cloud extent, incoming solar irradiance at the surface or precipitation.

3 The direct assimilation of radiances

The direct assimilation of satellite radiances is the most widely used method to assimilate satellite observations into NWP models. This technique is especially being performed and enhanced for operational GCMs by weather services around the world (BAUER et al., 2011b; GEER et al., 2017).

In the process of direct radiance assimilation, synthetic radiance observations are calculated from the model variables using a forward RTM as the observation operator (H in equation (2.1)). During the data assimilation step, the difference between these synthetic and observed radiances is considered. This difference is called an innovation. The impact on the observed and unobserved fields as a consequence of the combination of the innovation with all other assimilated observations, observation errors, background error covariance and covariance localization, is known as the analysis increment. This ultimately leads to the final analysis, i.e. the updated initial model state for the next model run.

The methodological advantage of radiance assimilation is that it makes direct use of the observed radiances or brightness temperatures (BT) without the need to first convert the observations into some retrieved property. Retrievals are generally more uncertain than radiances since they often rely on assumptions and auxiliary information, possibly from NWP models. The quantification of the radiance observation error is thus more accurate (MIGLIORINI, 2012).

3.1 Clear-sky radiance assimilation

Within the subject of the direct assimilation of radiances it is prudent to distinguish between cloud-free and cloud-affected radiance assimilation. While the assimilation of clear-sky radiances has already been performed for several decades in operational global models, it is still a recent topic in regional modeling. One reason for that is that the analysis is strongly influenced by the initial and lateral boundary conditions of the driving global model. Other reasons include difficulties associated with bias correction given the lack of global data and uncertainties associated with the land surface specification. The latter aspect is more important for regional-scale models given that they are typically located over land. Furthermore, global NWP models traditionally rely on polar-orbiting satellite sensors, which are going to be less useful for regional-scale models given their infrequent coverage of a limited-area domain.

Several recent studies evaluated the impact of clearsky radiance assimilation on cloud-related parameters, e.g. water vapour profile, cloud mask and precipitation. In the study of ZOU et al. (2011) and the follow-up study of QIN et al. (2013) coastal quantitative precipitation forecasts (QPF) were improved by assimilating GOES (Geostationary Operational Environmental Satellite) clear-sky IR radiances in the WRF model using the NCEP (National Centers for Environmental Prediction) GSI (Grid Point Statistical Interpolation) 3D-Var system (SHAO et al., 2016). The assimilation of the GOES observations results in a large added value compared to solely conventional data (like synoptic stations, radiosondes, aircraft reports and wind retrievals) in the study of Zou et al. (2011) and likewise in a positive impact in addition to the assimilation of multiple polar-orbiting satellite observations in QIN et al. (2013). ZOU and DA (2014) further developed the assimilation strategy and aim at a full utilisation of cloud-free radiance observations. They implemented a regime-dependent cloud mask for the removal of cloud-affected GOES radiance observations. The outcome is a more precise cloud mask determination when assessed using a MODIS (Moderateresolution Imaging Spectroradiometer) derived cloud mask. These studies prove that even without considering cloud-affected radiances, the assimilation of radiances observed in regions without clouds prior to convective initiation has a positive impact on short-term cloud and QPF forecasts.

A positive impact of clear-sky radiance assimilation on tropospheric moisture in WRF has also been found in experiments by SINGH et al. (2010, 2016) with the WRFDA (WRF model data assimilation system) 3D-Var system (BARKER et al., 2012) and model domains centred over India: The first-time assimilation of clear-sky water vapour-sensitive radiances of the Indian satellite Kalpana into WRF results in improved analyses and short-term forecasts, especially for mid-upper tropospheric moisture (SINGH et al., 2010). While the month-long simulations prove the benefits of this practice, extended experiments with more than one channel are desirable. Consequently, in the experiments of SINGH et al. (2016) several channels of the most recent Indian satellite INSAT-3D (Indian National Satellite System) have been considered for the first time. Assimilating INSAT-3D clear-sky temperature and water vapour-sensitive radiances results in improved tropospheric moisture and temperature profiles in the analysis and improved forecasts of moisture, wind, temperature and precipitation.

As YANG et al. (2017) demonstrate by applying a 3DEnVar approach with WRFDA, hybrid methods of variational and ensemble DA methods offer a great potential to improve convection-permitting LAM forecasts of clouds. The case study results over Mexico using GOES imager clear-sky radiances show analysis improvements in terms of temperature and humidity that ultimately lead to improved 24 h precipitation forecasts.

The results of these various studies with WRF are proof of the positive impact of clear-sky radiance assimilation on regional forecasts of cloud-related parameters. In order to avoid radiation from the surface, clear sky radiance assimilation is usually performed using channels whose sensitivity profiles peak in the middle and upper troposphere. To a large extend this explains the positive impact on temperature and humidity profiles in these altitudes, ultimately impacting clouds and precipitation. With a horizontal grid spacing of 25 and 30 km the experiments of SINGH et al. (2010, 2016) were performed with rather coarse horizontal resolutions of the LAM. In contrast, ZOU et al. (2011) and QIN et al. (2013) used 10 km while YANG et al. (2017) used 4 km. The impact of different domain sizes and resolutions on the assimilation outcome should be the subject of future work. Moreover, all of these studies present case studies. Exhaustive evaluations for several months have not been performed. Those would strengthen the significance of the results, especially regarding the use of new satellites like INSAT-3D. Long-term evaluations of clear-sky radiance assimilation with 4D-Var or ensemble DA systems are likewise missing in the literature. Such studies could bring forth important information about the dependency of clear-sky radiance DA on the lateral boundary conditions and different synoptic weather conditions. Nevertheless, avoiding the use of cloud-affected radiances and only assimilating clear-sky radiances is not comprehensive. While some authors focus exclusively on the impact of clear-sky radiance assimilation, most of the research community has focused on the assimilation of cloud-affected radiances given the large potential benefit that can be derived from these observations.

3.2 Cloud-affected radiance assimilation

3.2.1 General issues

Since the past decade, there have been more concentrated efforts to assimilate cloud-affected radiances into global NWP models (GEER et al., 2017). Processing clear-sky and cloud-affected radiances in a uniform way in global models is currently one of the top priorities in operational satellite data assimilation. Concerning limited-area models, cloud-affected radiance assimilation is also increasingly investigated. The assimilation of cloud-affected radiances is considerably more challenging in both global and limited-area models for several reasons, discussed by several authors (KOSTKA et al., 2014; POLKINGHORNE and VUKICEVIC 2011; SEAMAN et al., 2010; VUKICEVIC et al., 2006). The main aspects are:

1. The nonlinearity of moist processes is difficult to take into account.

The observation operators have to take into account water, which appears in all phases in the cloudy atmosphere, e.g. as vapour, liquid, mixedphase or glaciated. Radiative transfer processes involving clouds and precipitation are nonlinear, which conflicts with the fact that variational DA systems require linearized and adjoint forms of the forward operators during short time windows (BAUER et al., 2011a,b; OTKIN, 2010). Otherwise the convergence of the minimization of the cost function is not guaranteed (KOSTKA et al., 2014). ERRICO et al. (2000, 2007a) show that the cost function can become multimodal when moist processes are involved and that the minimization algorithm finds local minima of the cost function but not its global minimum. Consequently, more iterations are needed which increases the computational requirements and spurious noise can be introduced in the adjoint model which then leads to numerical instability (POLKINGHORNE and **VUKICEVIC**, 2011).

2. Detailed information about cloud microphysical variables and their uncertainties are required.

Clouds and precipitation are discontinuous in time and space and the associated processes and uncertainties are generally not well modelled by NWP models (BAUER et al., 2011b; VUKICEVIC et al., 2006), which poses problems for the forward operators that require precise profiles of cloud parameters (WENG, 2007). BENNARTZ and GREENWALD (2011) discuss radiative transfer issues related to this circumstance. The accurate description of cloud-resolving model background errors is a highly complex problem regarding the structure of the error covariance matrix (WENG, 2007) that is usually assumed to be isotropic and homogeneous and constant during the assimilation window for variational data assimilation (BAUER et al., 2011a).

3. Cloud location errors are difficult to handle.

Regarding clouds, four cases are possible in satellite data assimilation: (1) Both the model and satellite have clouds (2) the model has a cloud that is not observed by the satellite (3) the satellite observes a cloud while the model simulates a cloud-free atmosphere (4) or both are clear (Fig. 1(c) and (d)). This uncertainty is referred to as model cloud misplacement or location errors (KOSTKA et al., 2014; POLK-INGHORNE and VUKICEVIC, 2011). Location errors also cause problems in the adjoint calculation when observed clouds do not exist in the model. These aspects limit the ability of cloud-affected radiance assimilation in generating new model clouds (POLK-INGHORNE and VUKICEVIC, 2011; SEAMAN et al., 2010). As in the case of the nonlinearity problem, EnKFs are more robust than variational techniques concerning cloud location errors (OTKIN, 2010; ZU-PANSKI et al., 2011).

Another challenge with LAMs is the development of bias correction schemes, especially in the situation where both clear and cloudy sky observations are assimilated. Compared to global NWP models, bias correction is more challenging in LAMs because their small geographic extent makes it unlikely that they will capture a wide range of synoptic weather conditions. This is important because biases may be dependent on the prevailing weather and in the case of all-sky radiance assimilation will also be tied to different cloud types (OTKIN et al., 2018). This issue is crucial for polar-orbiting satellite sensors because they only observe the LAM domain two times each day, but it will be less of an issue for geostationary sensors since they provide complete domain coverage, likely with temporal resolutions < 15 min. Even so, the above issues still necessitate development of innovative bias correction methods suitable for application in LAMs.

The diverse issues that radiance assimilation faces explain the fact that up to the present day a large amount of cloud- and/or rain-affected radiance observations are not considered in operational data assimilation systems (BAUER et al., 2011b; GEER et al., 2017). The complex observation operators and the strongly increasing number of channels of new-generation satellite sensors make radiance assimilation computationally costly, which is another disadvantage regarding operational NWP (MIGLIORINI, 2012).

In the following subsections we investigate how recent studies have approached the above-mentioned problems and their implications for future work.

3.2.2 Observation operators for cloud-affected radiance DA

Several publications tackle the problem of the nonlinearity of moist processes by the implementation and testing of new observation operators for cloud-affected radiances. The assimilation strategies for cloudy radiances are adapted to the relevant NWP models and their available assimilation methods while the general goal is to overcome the problems related to cloudy radiance assimilation. Since variational and ensemble-based methods work differently regarding the handling of nonlinearity and other issues, it is important to consider the findings of the studies with respect to the applied DA method.

One early example of an observation operator for both cloud-free and cloudy situations in a LAM is the one that was developed for the 4D-Var DA system of the RAMS (Regional Atmospheric Modeling System) model RAMDAS (Regional Atmospheric Modeling Data Assimilation System). It has been developed and firstly evaluated by GREENWALD et al. (2002, 2004). The multi-scattering radiative transfer model computes synthetic radiances of GOES IR and VIS channels under all weather conditions. An important finding of the adjoint sensitivity analysis is that radiances of different channels are sensitive to different types of clouds in terms of their thickness or phase. The system has been further evaluated by VUKICEVIC et al. (2004, 2006) in case studies of stratus and multi-layered clouds without convection where a positive impact on analyses and short-term cloud forecasts could be found in both cases. Unfortunately the system has not been extensively evaluated for the case of complex and convective cloud situations or in other geographical regions using other satellites.

More fundamental work on cloud-affected radiance assimilation using a 4D-Var system has been done by STENGEL et al. (2009, 2010, 2013). The utilised HIRLAM (High Resolution Limited Area Model) model is the first LAM that assimilated SEVIRI (Spinning Enhanced Visible and InfraRed Imager) radiances using 4D-Var (DA scheme initially described by GUSTAFSSON et al., 2001 and LINDSKOG et al., 2001). The developed observation operator uses the RTTOV (Radiative Transfer for TOVS) RTM that is maintained by EUMETSAT and a simplified moist-physics scheme that was developed at ECMWF (European Centre for Medium-Range Weather Forecasts). The application of the new observation operator results in a reduction of the analysis errors of the total integrated water vapour and a forecast error reduction for geopotential height, humidity and wind direction at most model levels and especially in the upper troposphere. A shortcoming of the experiments is the relatively coarse grid spacing of 22 km. Further experiments should be performed in order to investigate the performance of the system at convection-permitting resolutions where the high spatial resolution of SEVIRI can be better utilised.

RAMS and HIRLAM are two prominent examples using new and functional observation operators for geostationary cloud-affected radiance assimilation for 4D-Var in LAMs in mid-latitudes. Thanks to the frequently available satellite observations 4D-Var takes into account how the atmospheric state evolves. The main achievement of the method that becomes evident in the studies is that especially tropospheric humidity increments are improved, which positively impacts cloud evolution after the model initialisation. However, the computational cost of the adjoint of the observation operator is high for a large number of observations. Moreover, using 4D-Var, the analysis and forecast quality largely depends on the capacity of the model to represent cloud processes. Especially in convective situations and at high resolutions the method reaches its limits, which is to be investigated by future studies. For example, the evaluation of tropical convective events would be an extreme performance test.

KOSTKA et al. (2014) criticise that not much work has been done regarding the assimilation of VIS and NIR radiances and list several advantages of VIS and NIR compared to IR radiances concerning cloud information content. They tackle this issue with the development of a new observation operator for NIR and VIS reflectances for the COSMO (Consortium for Smallscale Modeling) model and its LETKF (Local Ensemble Transform Kalman Filter) DA system. The kilometrescale ensemble data assimilation (KENDA) system for the COSMO model, developed by the German Meteorological Service (Deutscher Wetterdienst, DWD) constitutes an ensemble modelling system that includes the assimilation of geostationary satellite observations in a cloud-resolving model. The KENDA system is outlined by SCHRAFF et al. (2016) while further insight is given by HARNISCH et al. (2016) and SOMMER and WEISS-MANN (2014). In their simulations with the new NIR and VIS observation operator using Meteosat observations KOSTKA et al. (2014) obtain the best results by horizontally smoothing the observations and by applying a parallax correction, which accounts for the slant satellite viewing angle through the atmosphere. Nevertheless, the usage of VIS observations is constrained to the day-time, which limits their utility. Yet, the development of observation operators for VIS channels constitutes an important contribution to maximise the use of geostationary satellite observations.

3.2.3 Handling observation and model background errors

With an increasing cloud amount the observation and forward operator errors often acquire non-Gaussian error characteristics, which is not in line with the common DA assumption of normally distributed errors. Recent works using COSMO-KENDA attempt to account for this circumstance with new methods for error estimation and bias correction (HARNISCH et al., 2016; OTKIN et al., 2018). HARNISCH et al. (2016) apply a dynamic observation error estimate depending on the cloud impact with COSMO-KENDA – a technique that has previously been applied in the assimilation of microwave observations in a GCM (GEER and BAUER, 2011) and thus been transferred to infrared radiance assimilation in a LAM. The positive result is that the first-guess departure (observation minus first-guess) statistics become closer to a normal distribution. OTKIN et al. (2018) present a new bias correction approach based on a Taylor series polynomial expansion of the observation departures that is able to remove both linear and nonlinear conditional biases from all-sky satellite brightness temperatures. Passive monitoring experiments using BTs from two WV channels on SEVIRI showed that the bias in the observation departures was greatly reduced when using higher order Taylor series terms and bias predictors sensitive to clouds and water vapour. Although these studies present important proofs of concept to improve all-sky (cloudfree and cloud-affected) IR radiance assimilation with a highly-resolved LAM (2.8 km grid spacing), a longterm evaluation of the methods and an extension to more channels would be desirable.

Further studies with the RAMS model and cloudaffected GOES radiances have been performed, addressing the above-mentioned problems (2) and (3). POLK-INGHORNE et al. (2010) compute mean model background error statistics at cloud-resolving grid spacing of 4 km in order to further improve the RAMDAS 4D-Var assimilation system. They introduce a simple cloud mask and distinguish between clear-sky observations and those with either low or high clouds. This way, cloud location errors can be reduced during the assimilation by choosing only points with the same type of cloud in both the model background and the observations. Nevertheless, the cloud type agreement between observations and model is often poor due to the imperfect representation of the modelled cloud situation. POLKINGHORNE and VUKICEVIC (2011) further evaluate the impact of the established model background error statistics in a larger domain while SEAMAN et al. (2010) investigate the assimilation of cloud-affected IR radiances with a cloud-free model background of RAMS at a grid spacing of 6 km. The latter investigation also addresses the cloud location error problem. In the studied case a mid-level cloud is present in the observations (Fig. 3(c)) but not in the dry model background (Fig. 3(a)). The difficulty for the 4D-Var system is to generate a physically consistent state between temperature, humidity, wind and clouds. The authors find that the temperature and humidity profiles are modified in the correct direction after the assimilation of GOES radiances, bringing the model closer to the generation of the observed mid-level cloud (Fig. 3(b)). Nevertheless, the model does not necessarily assume a consistent and realistic cloud as a consequence of the assimilated radiances (Fig. 3(d)). The study of SEAMAN et al. (2010) is one of the rare studies that explicitly examines the influence of the domain size on the DA results. In the studied case, increasing the domain size changes the initial model state and the observed cloud covers a smaller proportion of the domain which does not necessarily lead to improved results. The impact of different LAM domain sizes needs to be studied with other models, at different resolutions and under various meteorological conditions.

The different RAMS case study experiments (POLK-INGHORNE et al., 2010; POLKINGHORNE and VUKICEVIC, 2011; SEAMAN et al., 2010) show a high sensitivity of the 4D-Var DA system to diverse parameters like assimilation window length, allowed difference of observed and modelled BT, background error decorrelation length, domain size and the studied case. These are important findings that stress the difficulty of correctly configuring a 4D-Var cloud-affected radiance DA scheme for a cloudresolving LAM. The works are a step forward in solving the cloud location error problem, but do not completely solve or circumvent it. The generation of model clouds with 4D-Var radiance assimilation in case of a cloudfree background is a major issue for which no comprehensive solution exists today.

3.2.4 Preparing for the assimilation of new geostationary satellite observations

On the side of EnKF systems, besides the above-cited works with the COSMO model, several Observing System Simulation Experiments (OSSE) have been performed with the WRF model and the DART (Data Assimilation Research Testbed) (ANDERSON et al., 2009) system (CINTINEO et al., 2016; JONES et al., 2013a, 2014, 2017, OTKIN 2010, 2012a,b). While JONES et al. (2017) evaluate the potential impact of geostationary hyperspectral observations, the other studies have in common that they aim at evaluating the impact of assimilating simulated GOES-R radiance observations at cloudresolving resolutions. OTKIN (2010) showed that the assimilation of both clear-sky and cloud-affected synthetic GOES-R ABI (Advanced Baseline Imager) IR brightness temperatures improved cloud analyses for a convective case across the central U.S. OTKIN (2012b) subsequently showed that the analysis and forecast accuracy are noticeably impacted by the choice of the covariance localization radius. Following these studies, the impact of WV-sensitive IR brightness temperatures was firstly evaluated by OTKIN (2012b) resulting in improved cloud and moisture analyses for a cool season weather event. CINTINEO et al. (2016) and JONES et al. (2013a, 2014)



Figure 3: Illustrations of RAMS 4D-Var radiance assimilation experiments over Nebraska on 2 November 2001 with a cloud-free RAMS background. a–b) Soundings of temperature (black) and dew point (grey) at North Platte, Nebraska observed at 1200 UTC (dashed lines) and simulated by RAMS at 1145 UTC (solid lines) before assimilation (a) and after assimilation of GOES Sounder channels 3, 4, 7 and 11 (b). c) GOES visible image of an altocumulus cloud extending between 4.2 and 4.7 km above mean sea level taken at 1445 UTC. d) Model location and height of cloud top in km above ground level (AGL) after assimilation of GOES Sounder channels 7 and 11 with decorrelation lengths doubled revealing the largest cloud (along the Nebraska-Colorado border) from any experiment that may be considered 'mid-level' with a top height of 2.5 km AGL. Original figures adapted from SEAMAN et al. (2010). © 2010 The Authors. Published by Taylor & Francis. Used with permission.

assimilate both synthetic GOES-R BT and radar observations, showing that the two complement each other and that the satellite observations are vital. Distinct improvements of analysis and forecast accuracy and the simulated cloud field can be found throughout the experiments thanks to the assimilated synthetic satellite observations. The works constitute an important preparation for the usage of real GOES-R ABI observations with state-of-the-art cloud-resolving ensemble DA modelling systems. Since GOES-R is now operational as the GOES-16 satellite, the system should be thoroughly evaluated for real cases.

Besides the diverse works with GOES and SEVIRI observations, efforts to assimilate Himawari-8 all-sky radiance observations at cloud-resolving resolutions are also being made. OKAMOTO (2017) performed the first evaluation of simulated Himawari-8 all-sky IR radiances from four channels with the JMA (Japan Meteorological Agency) non-hydrostatic model (JMA-NHM). OKAMOTO (2017) suggests beginning with the assimilation of only WV-sensitive channels for further developments. Moreover, it is emphasised that the successful assimilation of cloud-affected IR radiances requires improvements of both the NWP model and the RTM. The

diverse problems (1–3, Section 3.2.1) are thus left unsolved once more. This is one of the few studies on radiance assimilation that compares the performance of the two most widely used observation operators RTTOV and Community Radiative Transfer Model (CRTM), finding a slightly better performance for the CRTM. However, this finding is to be confirmed in actual assimilation experiments with an ensemble-based or variational DA system.

3.2.5 Alternative methods

Alternative assimilation methods to 3D-Var, 4D-Var or EnKF have been tested by RAYMOND et al. (2004) and ZUPANSKI et al. (2011). The former study assimilated GOES brightness temperatures from a channel sensitive to upper-tropospheric water vapour by iteratively modifying the upper-tropospheric humidity in the CRAS (CIMSS (Cooperative Institute for Meteorological Satellite Studies) Regional Assimilation System) model at 40 km grid spacing. ZUPANSKI et al. (2011) perform a WRF case study with a grid spacing of 15 km of an extratropical cyclone over Europe with a maximum likelihood ensemble filter (MLEF) and the assimilation of synthetic all-sky IR GOES-R radiances of a window channel. It is an ambitious case study because of strong cloud location errors. Both studies are successful proofs of concept showing positive impacts of the assimilation on cloud-related parameters like upper-tropospheric moisture (RAYMOND et al., 2004) and cloud ice (ZUPAN-SKI et al., 2011). Extended studies of the methodologies at cloud-resolving scales, over a longer evaluation period, with more channels and real observations from GOES-R or SEVIRI or synthetic observations of Meteosat Third Generation (MTG) would strengthen the proofs of concept.

3.2.6 Concluding remarks

In summary, radiance assimilation is the common approach for satellite data assimilation adapting the NWP model's representation of the atmosphere to the satellite's view of the atmosphere. All-sky IR radiance assimilation holds a lot of promise but has not traditionally been performed with LAMs because of the many challenges associated with it. Diverse problems occurring in cloudy situations are difficult to handle and solutions for these problems are being developed and tested. Meanwhile, radiance assimilation proves to be very beneficial for cloudiness forecasting, even if cloud-affected observations are not considered. Some important proofs of concept have been performed that still have to be supported by long-term evaluations. In particular, the topic of cloud-affected radiance assimilation in LAMs at cloud-resolving resolutions has received limited attention so far. Since 4D-Var systems struggle with accurate estimates of the model background error and the capacity of the models to simulate clouds, convectionpermitting simulations with these systems are challenging and require a thorough adjustment of the modelling and DA system. Especially in the case of a cloud-free model background when clouds have to be entirely generated, the 4D-Var approach reaches its limits as a result of the unknown formulation of a balanced B-matrix for cloud parameters. This is less critical for EnKFs which would draw more attention to members already containing clouds. EnKFs are thus more robust and allow to focus more directly on the assimilation of cloud-affected radiances in combination with the general improvement of LAMs and forward operators. So far, to our knowledge, there is no peer-reviewed article that evaluates hybrid (meaning both variational and ensemble) DA methods for all-sky radiance assimilation with LAMs.

4 The assimilation of physical cloud properties

Observed radiances of geostationary meteorological satellites are routinely converted into various meteorological quantities, such as physical cloud properties. In this section we focus on the assimilation of cloud property retrievals since they are expected to have the largest impact on cloudiness forecasts. Physical cloud properties derived from geostationary satellite observations include cloud cover, cloud top pressure (CTP), cloud top height (CTH), cloud top temperature (CTT), liquid water path (LWP), ice water path (IWP), singlelayer or double-layer cloud amount, effective cloud amount (ECA), cloud optical depth, phase and effective particle size (BAYLER et al., 2000; DERRIEN and LE GLÉAU, 2005; JONES et al., 2013b; MINNIS et al., 2008; WATTS et al., 2011; YUCEL et al., 2002). The goal of cloud property assimilation is to convert the satellite observations into vertical distributions of cloud water and ice for each model column (YUCEL et al., 2002).

Cloud property assimilation can be advantageous compared to radiance assimilation. For example, most satellite-derived cloud properties can be directly compared to the properties derived by NWP models. This avoids the computationally costly and complex application of RTMs in the data assimilation procedures (JoNES et al., 2013b) and makes the assimilation process more independent of a given satellite.

4.1 Historical overview

In the last two decades, various approaches aimed at deriving atmospheric analyses for limited-area models with more realistic cloud presence estimations, using geostationary satellite observations. The first works on this subject appeared in the 1990s, e.g. with the works of (LIPTON and VONDER HAAR 1990a,b). They compared several analysis methods using retrievals of atmospheric temperature and water vapour mixing ratio profiles and surface temperatures, retrieved by the VISSR (Visible and Infrared Spin Scan Radiometer) Atmospheric Sounder (VAS) on GOES and assimilated the observations using a model first-guess of the RAMS model. The applicability of the method was proven in a two-dimensional simulation of a vertical cross section in a mountainous region (LIPTON and VONDER HAAR, 1990b) and a summertime case study (LIPTON and VON-DER HAAR, 1990a).

Afterwards, studies using cloud shading retrievals were performed. LIPTON (1993) and MCNIDER et al. (1995) made use of GOES visible observations in cloud shading assimilation experiments where they analysed the impacts of the modified surface temperature due to cloud shading on the planetary boundary layer and cloud development. The method of McNIDER et al. (1995) was later used by several other authors for land surface and air quality related experiments.

Also in the 1990s, MACPHERSON et al. (1996) applied the DA system MOPS (Moisture Observation Preprocessing System) and the UKMO (UK Met Office) mesoscale model, testing the assimilation of Meteosat IR imagery together with other sources of observations. The observations were used to generate a three-dimensional analysis of cloud fraction that has been converted into synthetic humidity profiles that are assimilated the same way as radiosondes. The concept of generating a threedimensional cloud analysis from several observation sources had also been applied operationally with the limited-area model NAE (North Atlantic/European) using a 4D-Var DA system (RENSHAW and FRANCIS, 2011; TAYLOR et al., 2008).

Another often-cited cloud analysis scheme that makes use of several sources of observations is the ARPS (Advanced Regional Prediction System) Data Analysis System (ADAS). Souro et al. (2003) present a historical overview of the development of ADAS, whose cloud initialization procedure is an advanced version of NOAA's (National Oceanic and Atmospheric Administration) LAPS (Local Analysis and Prediction System) system (described by ALBERS et al., 1996). Numerous physical cloud properties are implemented in the analysis package, which enables the determination of a threedimensional cloud and precipitation analysis.

4.2 Modern methods

The historical overview shows that many early studies on cloud property assimilation are still relevant since the concepts are often transferred to new models. Common conceptions and difficulties of state-of-the-art methods are examined in the following, and particularly important works are highlighted.

4.2.1 Handling the four possible clear-cloudy cases

As mentioned earlier, four cases can be distinguished in all-sky DA, depending on whether cloud presence is satellite-observed (yes/no) and model-simulated (yes/no). Several recent studies considered these four cases and designed the cloud analysis scheme in a way that it can handle each case. Evidently, the case in which clouds are observed by the satellite but not simulated by the model is the most difficult one since the complete three-dimensional cloud characterisation has to be created in the model. Generally, cloud property assimilation offers more possibilities to improve the cloud analysis than radiance assimilation, since it allows a more direct and distinct modification of modelled cloud fields.

YUCEL et al. (2002) developed an innovative cloud analysis scheme for the RAMS model that handles the four cases. The scheme makes use of GOES VIS imager derived estimates of the vertically integrated cloud water/ice. The critical case in which clouds are not simulated by the model but observed by the satellite is solved by incrementally increasing the whole-column mass of cloud water until it equals the satellite observation. Positive impacts of this cloud injection procedure on shortterm forecasts of downward surface shortwave and longwave radiation and cloud cover are found. A major disadvantage of the method is that it is based on VIS observations and thus not suitable for night-time simulations. YUCEL et al. (2003) accounted for this by including IR derived cloud-top BT and height observations in the cloud injection algorithm. The improved method was tested in a case study with the fifth-generation Pennsylvania State University–National Center for Atmospheric Research (PSU–NCAR) Mesoscale Model (MM5) at 4 km grid spacing. A positive impact on short-term forecasts of downward surface solar irradiance and precipitation can be found. The biggest problem is that this positive impact rapidly decreases with increasing forecast lead time due to the mismatch between the updated model cloud cover and the unchanged vertical wind speed field. The problem of finding a consistent atmospheric state in the cloud analysis scheme has been tackled in some studies which are mentioned in the next subsection.

4.2.2 Assimilating cloud-top information

As in the works of YUCEL et al. (2002, 2003), in many recently developed cloud property assimilation methods that have been applied to diverse LAMs the focus is mostly set on the assimilation of cloud-top information, e.g. cloud top temperature or pressure/height (DE HAAN and VAN DER VEEN, 2014; GUIDARD et al., 2006; MATHIESEN et al., 2013; RENSHAW and FRAN-CIS, 2011; SCHOMBURG et al., 2014; TAYLOR et al., 2008; VAN DER VEEN, 2013; WHITE et al., 2018), since optical and thermal sensors are not able to directly capture information inside clouds. As satellite-derived cloud properties may have large errors, all methods have in common that the vertical position, extent, and number of cloud layers may not represent the truth. Some authors make use of other sources of observations (e.g. cloud base height or radiosondes) to better account for this limitation. Using additional observations is obvious in regions where several types of observations and a dense ground-based observation network are available. Nevertheless, this means that the respective methods are not globally applicable. In regions where geostationary satellites are the only considerable source of observations, more sophisticated methods to derive the vertical properties of clouds are needed.

One technique for the determination of the cloudtop height of the modelled clouds has been applied by several authors: In the DA methods of YUCEL et al. (2003), GUIDARD et al. (2006), MATHIESEN et al. (2013) and VAN DER VEEN (2013) the CTH is defined by the model layer whose temperature equals the satellitederived CTT. This is a suitable approach since satellitederived cloud-top information are rather accurate in the presence of optically-thick clouds and no additional information than the satellite observation is necessary. The shortcoming of this method is that it implies that the simulated vertical temperature profile is correct, which is not necessarily the case. SCHOMBURG et al. (2014) use a different approach to determine the model cloud top and find a compromise between observation and model. In their approach the layer that is at the same time close to the observation and close to saturation is found by minimising a cost function. Since ensemble DA is used in their study, the ensemble members that are closer to the observation get a higher weight. Further approaches are imaginable to optimize the CTH assignment in the cloud analysis. For example, future methods might consider uncertainty information about CTH provided by an ensemble, considering all ensemble members.

4.2.3 The modification of vertical profiles

A common technique to initialise clouds is the direct modification of modelled profiles and fields according to the observed retrievals, e.g. profiles of humidity or water vapour, temperature and/or liquid/ice water content (MATHIESEN et al., 2013; VAN DER VEEN, 2013; YUCEL et al., 2002). The technique is risky since it might introduce numerical instability. That is because the entirety of atmospheric fields is not coherent anymore after the cloud analysis - A problem which can be reduced by horizontal and vertical smoothing (MATHIE-SEN et al., 2013) or digital filtering (VAN DER VEEN, 2013). This circumstance shows that cloud property assimilation is more experimental than radiance assimilation. Most cloud property assimilation methods focus on vertical columns at the analysis time rather than including the temporal and three-dimensional evolution of cloud properties. This is another major general point of criticism that should be approached with greater detail by future methods. Some methods which are oriented towards that direction are discussed in Subsection 4.2.7.

4.2.4 Impacts on the wind field analysis

As mentioned before, a major conceptual difficulty of cloud property assimilation is that the focus is often exclusively set on cloud-related atmospheric parameters like humidity, without changing the initial threedimensional wind field of the model (GUIDARD et al., 2006; LAUWAET et al., 2011; MATHIESEN et al., 2013; TAYLOR et al., 2008; VAN DER VEEN, 2013; YUCEL et al., 2002). Hence, the cloud analysis might be more accurate after the assimilation but the short-term forecast does not necessarily improve due to the unsuitable wind field. It can be assumed that this problem becomes more important with increasing horizontal resolution and at cloud-resolving scales. This circumstance has not yet been analysed in peer-reviewed literature. Introducing a cloud into the analysis where there was none before might subsequently change the dynamic fields (e.g. the wind field). Therefore, CHEN et al. (2015) suggest to analyse the interactions between cloud microphysics and the model dynamics in order to further enhance cloud analysis schemes. A nudging approach which tackles the problem entirely from the dynamical point of view has been tested by WHITE et al. (2018). In their method that has been applied to WRF, CTT and cloud albedo observations are used to modify the vertical velocity and divergence fields of the WRF analysis. The changes of vertical velocity have a direct impact on convection and thus the generation and dissipation of clouds. Nevertheless, since a purely dynamic approach is used, this method is not capable of considering the vertical cloud structure and cloud type or multi-layered clouds. Future cloud analysis methods based on cloud products should take account of the dynamical response of an "injected" cloud analysis.

Wind field adjustments could be provided from a different observing system such as radar, atmospheric motion vectors (AMVs) or cloud motion vectors (CMVs). A review of the assimilation of these properties is beyond the scope of this paper. Nevertheless, to our knowledge geostationary-observed AMVs or CMVs have not yet been used within geostationary-observed cloud property assimilation schemes for LAMs. This is a potential future field of research.

4.2.5 Multi-layered cloud analyses and the use of multiple cloud properties

Cloud property assimilation methods generally manage to place clouds at the correct locations, according to the satellite observations, but not necessarily their true characteristics and vertical properties. For example, many methods do not favour the development of multi-layered clouds in the NWP model (GUIDARD et al., 2006; KU-MAR and VARMA, 2016; MATHIESEN et al., 2013; TAY-LOR et al., 2008; VAN DER VEEN, 2013). The method presented by BAYLER et al. (2000) is innovative since it allows multi-layered clouds in the cloud analysis, which is not the case for many later works. In the case that a low cloud exists in the model and the observed cloud top is considerably higher, the original cloud is left unchanged and an additional higher cloud layer is added. Besides GOES-derived CTP the method based on a successive corrections algorithm also uses an ECA product. The shortcoming of their algorithm is that it does not take into account the cloud type but only cloud presence. The principle of the method might be revived and improved by making use of multi-layered cloud products.

Integrating multi-layered cloud products to determine a cloud analysis is one potential strength of cloud property assimilation. MINNIS et al. (2008) have developed multi-layered and nearly global cloud products derived from multiple geostationary satellites and named Global Geostationary Gridded Cloud (G3C) products. CHEN et al. (2015, 2016), whose method is discussed in Section 4.2.7, are the only ones who made use of the G3C products in a LAM so far. However, they did not make use of multi-layer information. A sophisticated use of multi-layer cloud properties could open new opportunities for cloud property assimilation and should be considered in future methods.

Multi-layered clouds are explicitly excluded in a cloud analysis method for WRF named Cloud Data Assimilation (CLDDA) (MATHIESEN et al., 2013). The method that aims at improving solar irradiance fore-casts focusses on coastal stratocumulus clouds in California which are expected to have a rather stable thick-ness. It makes use of GOES-derived CTT and determines the cloud-base height using an empirical calculation. WRF-CLDDA has been further tested by YANG and KLEISSL (2016) who applied CLDDA in combination with a preprocessing scheme that uses additional NWP input. WRF-CLDDA significantly improves short-term solar irradiance forecasts in California in cases with a strong influence of stratocumulus clouds in both studies. SCHIPPER and MATHIESEN (2015) adapted the cloud analysis scheme to the Meteosat-based MPEFs (Meteorological Products Extraction Facility) Optimal Cloud Analysis (OCA) product. They attest its positive impact on solar power forecasts in a comparably long evaluation time of eight months. In summary, CLDDA is a simple and easily adaptable cloud analysis scheme which might be further improved for multi-layer clouds and different cloud types and tested with cloud products of other satellites, in different geographical regions.

An innovative system that makes use of radiances from multiple polar-orbiting and geostationary satellites to detect multi-layer clouds and retrieve their vertical extent is the Multivariate Minimum Residual (MMR) scheme (AULIGNÉ 2014a,b; DESCOMBES et al., 2014; XU et al., 2015). The scheme derives the cloud fraction for each vertical model level using observed and modelderived radiances. It has been successfully applied to the global model of ECMWF (the Integrated Forecasting System (IFS)) and to WRF. The good performance of the system can especially be achieved thanks to the intensive use of polar-orbiting satellite observations with many channels. A comparison with the other methods and studies discussed in this paper which focus only on geostationary satellite observations, is therefore inappropriate here. Nevertheless, the ability of the system to derive a multi-layer cloud analysis is remarkable and the method will likely influence future developments in the field of cloud analysis determination, particularly concerning new geostationary satellites with more channels.

4.2.6 The use of multiple cloud products

Most recently developed cloud property assimilation methods use a relatively small selection of retrieved cloud properties, but not the majority of available properties at the same time to create an as complete cloud analysis as possible. The method of LAUWAET et al. (2011) for example is one of the rare methods that make use of cloud optical thickness, but no additional cloud properties. The focus on a specific choice of properties is due to the fact that the chosen ones have to be compatible with the model variables and the capabilities of the DA system. For example, JONES et al. (2013b) describe that assimilating both GOES cloud water path and cloud ice path is challenging for deep, multiphase clouds due to the detailed distinction of the cloud phase in the LAM and the binary distinction between liquid or ice of the satellite. Other factors like uncertainty treatment, quality control and computational efficiency also influence the choice of cloud properties to be used in the assimilation method (ERRICO et al., 2007a).

Besides, for certain regions of the earth the cloud products of multiple geostationary satellites over the same region might be used. Over the Indian Ocean and India for example, three geostationary satellites (INSAT-3D, Meteosat-8 and Kalpana) could currently be used in order to derive a comprehensive cloud analysis.

4.2.7 Variational and EnKF methods

Many of the previously mentioned publications more or less forcefully inject or remove clouds in the LAM analysis for deterministic forecasts and thus constitute computationally fast alternatives to variational or EnKF methods. Besides these developments, a few authors focus on the development of observation operators for variational or EnKF DA systems that make use of retrieved cloud properties.

One example for cloud property assimilation with an EnKF system at the convective scale (2.8 km grid spacing) is the work of SCHOMBURG et al. (2014) who present a new method to assimilate SEVIRI-derived cloud mask, cloud classification and CTH information. The actual variables that are assimilated to the COSMO ensemble are derived pseudo observations of CTH and relative humidity. Tested for single-observation experiments, the method leads to improved profiles of cloud-related variables. Non-assimilated variables (e.g. temperature and wind) are modified through cross-correlations of the background ensemble. Besides the ideas in the previous sub-sections to further improve the methodology, an extension of the method to assimilate more than cloud-top information, e.g. multi-layer cloud properties, might be done in the future. Moreover, the method might be put to the test for convective summertime cases and the impact of the assimilation on the wind field might be analysed.

JONES et al. (2013b) are the first to assimilate cloud water path (CWP) at convection-permitting resolution (3 km grid spacing). A forward operator for the assimilation of CWP with an EnKF (WRF-DART) is evaluated in this work. At the same time the study focusses on cloud property assimilation with a LAM using an EnKF. The application of the new forward operator in a case with pronounced convection over the continental US shows an improved analysis of shortwave downward solar irradiance. Fig. 4 illustrates the impact that this kind of cloud property assimilation can have on the analysis. The CWP difference plots of conventional observation assimilation (CONV) and conventional plus CWP assimilation (PATH) reveal the ability of the method to both remove and create clouds in the LAM and thus correct for cloud location and extent. While variational radiance DA without the use of an ensemble has difficulty to produce entirely new model clouds (Fig. 3, SEAMAN et al., 2010), the assimilation of CWP with an EnKF proves to introduce new clouds effectively. Nevertheless, a thorough comparison of the two methods would require their application to the same weather situation. This would allow to examine the limits and differences between purely variational and purely ensemble-based DA as well as pure radiance and pure cloud property



Figure 4: Difference in WRF 40 member ensemble mean CWP over Oklahoma and neighbouring states after assimilation of conventional observations (CONV) and GOES CWP observations in addition to the same conventional observations (PATH) at 1945 UTC (a) and 2045 UTC (b) on 10 May 2010. Blue regions indicate where PATH generates lower values of CWP than CONV whereas red regions indicate that PATH generates higher CWP values. Original figure reprinted from JONES et al. (2013b). ©American Meteorological Society. Used with permission.

assimilation. After the successful proof of concept and a follow-up case study by JONES et al. (2015) that includes radar DA in addition to the CWP DA, an evaluation over a longer period and under various conditions is needed to strengthen the verification results for this method.

CWP has also been assimilated by CHEN et al. (2015, 2016) who are the first to do this with a 3D-Var system in a LAM. Their forward operator for cloud ice water path and cloud liquid water path for WRFDA has been tested over a 10 day period with a model grid spacing of 12 km (CHEN et al., 2015). Positive impacts on diverse variables, especially in the lower stratosphere are found. A subsequent study investigated the performance of the method in combination with different microphysics schemes with a 12 km/4 km two-way nest setup (CHEN et al., 2016). The authors stress that in order to further improve cloud property assimilation with a 3D-Var system, the background error covariance for hydrometeors should be anisotropic, inhomogeneous and flow-dependent. This could be achieved with a hybrid DA method using background information provided by an ensemble. The experiments of JONES et al. (2013b, 2015) and CHEN et al. (2015, 2016) for CWP assimilation might be extended to the use of multi-layer cloud information. Besides, the authors list several ideas to further improve the forward operators. Future methods might take up these examples and try to make better use of multiple derived cloud properties and newly available cloud properties (for example from GOES-R, Himawari-8 and Meteosat Third Generation) to more accurately map the observed cloud situation into the LAM.

An innovative work that performs rainfall retrieval assimilation with a 4D-Var system (WRFDA) is presented by KUMAR and VARMA (2016). They are also

the first to our knowledge to assimilate a geostationary satellite-based rainfall product (INSAT-3D Hydro-Estimator rainfall) with a LAM. Assimilating rainfall with 4D-Var is problematic since precipitation rate errors are not normally distributed. The advantage is that there is no need to develop a new sophisticated observation operator for the satellite product as modern 4D-Var systems are already able to assimilate conventional rainfall observations. The method has been tested for a summer monsoon case study and proves to improve shortterm rainfall forecasts. The authors suggest that an improved model background would lead to better results. One reason for this, among others, is that less observations would be rejected in quality control when the difference to the first guess is too large. An idea that has not been exploited so far is to make use of the high temporal availability of cloud property observations in 4D-Var over the duration of the assimilation window. The works of KUMAR and VARMA (2016) and CHEN et al. (2015, 2016) can be considered fundamental for the achievement of this goal. Future studies can build upon this work and aim at making a more sophisticated use of modern cloud products in 4D-Var DA.

4.2.8 Concluding remarks

In summary, there is a huge variety of cloud property assimilation methods that all have their specific strengths and deficits. Two general strategies can be distinguished: "forced" cloud injection mostly via the modification of cloud-related variables in vertical columns and computationally much more expensive variational or ensemblebased methods that are technically closer to radiance assimilation. This variety, together with the fact that the method always has to be tailored to the applied model, makes it practically impossible to determine the most effective method. Comparisons of different models and methods under identical conditions would be necessary in order to measure the individual impact of the methods. This, in turn, would require standardised evaluation procedures for cloud property assimilation.

5 Discussion

After the critical overview of the particularities of radiance and cloud property assimilation methods in the previous sections, this section discusses comprehensive aspects of geostationary satellite data assimilation in LAMs for short-term cloudiness forecasting. Moreover, the differences between satellite DA in research and the operational application and future perspectives are discussed.

BAUER et al. (2011a) and VUKICEVIC et al. (2006) remark that NWP models produce spatial and statistical averages of clouds while the information content about clouds provided by satellite observations is of a different nature. This circumstance explains the complexity of determining four-dimensional cloud analyses and why many authors focus on moisture analyses rather than the adaptation of variables linked with hydrometeors. It is this issue on the side of the weather models and the complete exploitation of available cloud observations on the side of remote sensing which still offer much room for improvements.

Several studies in both fields of radiance and cloud property assimilation raise the crucial question how long the assimilated information provided by satellite observations is influential in the model forecast. The answer to this question illustrates the impact of satellite data assimilation, which can be considerably large. Several works (BAUER et al., 2011a,b; BAYLER et al., 2000) hypothesize that the additional information by cloudaffected satellite data vanishes after the first few hours of simulation. YUCEL et al. (2003) assume that this is because of potential discrepancies between the unchanged dynamic field and the updated cloud analysis. While a loss of cloud information within the first 24 hours is found in several publications (CHEN et al., 2015, 2016; LIPTON and MODICA, 1999; SINGH et al., 2010; TAY-LOR et al., 2008), there are also multiple examples in which regional forecasts improve for lead times beyond 24 hours thanks to geostationary satellite data assimilation (VAN DER VEEN, 2013; ZAPOTOCNY et al., 2005; Zou et al., 2011). This proves that the impact can potentially be large and that further improvement of the methods might lead to longer lasting impacts. Nevertheless, the predictability of clouds depends on their type. Due to their limited predictability, convective clouds cannot be expected to display long impact durations when assimilated. HOHENEGGER and SCHÄR, (2007) demonstrate that the high degree of nonlinearity in cloud-resolving models limits predictability at cloud-resolving scales.

Many studies evaluate the functionality of their new developments in case studies, which is probably due to limited computational capacities, particularly regarding ensemble-based methods. Apart from the satellite DA method, the configuration and tuning of the model also plays a big role regarding the impact duration, ensuring that the injected clouds do not directly disappear after the initialisation. For example, CHEN et al. (2016) found that among several tested microphysics schemes the use of the WRF Double Moment 6-class (WDM6) scheme leads to the longest impact duration in their case. In general, more studies should evaluate the applied DA methods in connection with the model configuration in order to optimise the forecast impact duration. The LAM domain size certainly plays an important role regarding the impact duration, since the information given from the global to the limited-area model via the lateral boundary conditions highly influences and "washes out" the initial information in the regional domain at some point (GUSTAFSSON et al., 2018). We suppose that this effect becomes smaller the larger the domain is, as some studies mentioned in the previous sections indicate. This is to be validated in detail by future studies.

Moreover, to our knowledge no peer-reviewed paper compares the impact duration of radiance and cloud property assimilation in LAMs. Both strategies are able to produce positive impacts but a definite decision regarding which one has the larger impact on short-term forecasts of clouds cannot be made at this time.

The majority of the studies focuses on the assimilation of few channels or cloud properties in order to achieve a positive impact. Thus, the entirety of available channels and cloud properties is still far from being exploited. This is because for a respective NWP and DA system, diverse issues like bias correction, quality control or error estimation have to be thoroughly handled before extending the number of utilised observations.

Moreover, physical cloud property DA is linked to the errors induced by cloud property retrieval algorithms, additionally to the error of the observed radiances. Since the initial observations are radiances, the information content about cloud and precipitation physics is limited (ERRICO et al., 2007b) and sometimes biased, especially when optically thin clouds create a mixed signal of radiation from the background and the cloud (POLKINGHORNE and VUKICEVIC, 2011). Such problems can be overcome by using NWP output to correct the cloud information. Therefore, in the data assimilation scheme of the same model, the same background information is used twice, which might lead to suboptimal analyses (MIGLIORINI et al., 2008). The use of NWP background information in retrieval algorithms is thus a major shortcoming of cloud retrieval assimilation.

According to the majority of studies in Section 3, radiance assimilation especially improves uppertropospheric moisture and cloud features in situations with "smooth" cloud systems like stratus clouds, but has less attested impact on convective clouds and their evolution. Cloud property assimilation methods allow to inject clouds more radically and thus have more potential to improve short-term forecasts in convective situations. Unfortunately, only a few cloud property assimilation studies consider ensembles. Important uncertainty information might be derived from the ensemble and further improve the methods. In this regard, the implementation of stochastic elements in parameterization schemes might provide enhanced information about the ensemble spread (BENGTSSON et al., 2013). Furthermore, there is a lack of extensive 4D-Var studies of radiance assimilation with the widely-applied WRF model. These could be compared to the diverse ensemble-based radiance DA developments.

For these various reasons, the realisation of longterm comparison studies is desirable in order to fully evaluate the potential of the innovations and the forecast impact duration. Moreover, long-term studies of both radiance and cloud property assimilation methods with the same model would allow a more thorough assessment of the impact of the approaches with a limited influence of the model configuration and tuning. The complexity of cloud property assimilation methods is diverse (DE HAAN and VAN DER VEEN, 2014; SCHOMBURG et al., 2014; WHITE et al., 2018). Some methods are tailored for certain cloud types (MATHIESEN et al., 2013) or cloud properties (JONES et al., 2013b; KUMAR and VARMA, 2016). Comparing the performance of different methods under similar conditions would reveal whether more complex or computationally expensive methods perform better than simpler methods. Nevertheless, the conduction of such studies requires a common and globally applicable verification framework to assess the impact of the different methods. Such a framework does not currently exist and every author group uses different data for cloud analysis and forecast evaluation, like polar-orbiting satellite observations, reanalyses, radar, irradiance or other ground-based observations.

The evolution of clouds is far from being exhaustively considered in the process of determining a cloud analysis in both radiance and cloud property assimilation. Smoothing data assimilation methods (VAN LEEUWEN, 2001) aim at improving the temporal consistency and have not yet been extensively applied to the diverse cloud analysis issues. The shortcoming is that they are computationally expensive and require four dimensional adjoint models (VUKICEVIC et al., 2006). Nevertheless, it is desirable for both radiance and cloud property assimilation to better account for observed cloud evolution in the assimilation process.

Differently than in research, operational activities require a definite choice of the applied data assimilation procedure. Furthermore, while the impact of different satellite data assimilation methods on cloudiness forecasts can be investigated in detail in a research context, the operational context requires decisions to be made. The choice of an operational assimilation method always depends on the chosen model and its available data assimilation capabilities. The method has to be customised for the applied NWP model and its state or moisture variables. It is rather the entirety of available observations of all types that is considered in an operational context as well as the improvement of scores of all forecasted variables in the entire three-dimensional model domain. By trying to assimilate a maximum of available observations, the borders between radiance assimilation and cloud property assimilation become blurred. A comparison of research and operational activities and achievements reveals that there is a transition going on in operational cloud analysis determination and short-term forecasting of cloud-related parameters: Clear-sky radiance assimilation using cloud mask/cloud type classifications and the assimilation of simple cloud information like cloud-top temperature can be considered the operational standard with LAMs (GUSTAFS-SON et al., 2018). This is now being updated by a more sophisticated assimilation of elaborated cloud products and all-sky radiances.

Several systems – especially operational ones – use multiple observations to make an exhaustive threedimensional cloud analysis. Previously mentioned examples are the systems or models ARPS, LAPS, MMR and NAE. Two operational state-of-the-art systems that have not been mentioned yet are the Rapid Refresh (RAP) model and the AROME (Application de la Recherche à l'Opérationnel à Méso-Échelle) model.

For short-term forecasting with a forecast length of 18 hours NCEP operationally runs RAP, which replaced the Rapid Update Cycle (RUC) model in 2012 (NOAA NCEI 2017). Rapid Refresh is an assimilation/modelling system, based on RUC, WRF and GSI (BENJAMIN et al., 2016b) that uses a hybrid 3D-Var/Ensemble data assimilation approach. GOES-derived cloud-top height temperature are incorporated as it had already been the case in the RUC system (BENJAMIN et al., 2004). Hu et al. (2007) and WEYGANDT et al. (2006) provide further information about the cloud analysis scheme of RUC/RAP. Since August 2016 the assimilation system also integrates GOES radiances (BENJAMIN et al., 2016a).

In the AROME model, operated by Météo-France, SEVIRI clear-sky radiances are selected using a cloud type product and assimilated with a 3D-Var system (GUIDARD and FOURRIÉ, 2010) in an hourly assimilation cycle (BROUSSEAU et al., 2016). The model is highly resolved with a grid spacing of down to 1.3 km and the impact of using a higher resolution on convective events in conjunction with DA has been analysed by the latter authors.

Unfortunately, detailed evaluations of the impact of only geostationary satellite observations in such operational systems are usually not available and the systems are tailor-made for certain geographical regions and the respective observations.

6 Conclusions

In this paper we present a comprehensive review about the assimilation of atmospheric observations made by geostationary meteorological satellites in limited-area models with the goal of improving short-term forecasts of cloud-related parameters.

Two fundamentally different approaches can be distinguished: sensor-observed radiance or brightness temperature assimilation and retrieved cloud property assimilation. The question of which assimilation strategy is the most suitable one for a certain application depends on several factors: the available computational capacity, the requirements of the performed simulations in terms of forecasted variables and computational cost and the sophistication of the NWP model with its associated DA system. Generally, cloud property assimilation is rather targeting computationally spare and frequently updated forecasts of short-term cloud evolution, while radiance assimilation is usually applied in connection with computationally costly variational or ensemble-based DA systems which are designed to incorporate various sources of observations for short-term and long-term forecasts of not only cloud-related parameters. It is thus primarily a matter of lead time and application which technique should be focussed on, while in the end both methods aim at finding optimal NWP initial conditions.

The review shows that in both fields, radiance and cloud property assimilation, numerous improvements have been made recently. All radiance assimilation studies have found a positive impact on analyses and/or forecasts of cloud-related parameters through assimilation of geostationary satellite radiances. Reduced errors are mainly found for moisture and temperature fields especially in the mid-to-upper troposphere and for cloud-related parameters (e.g. cloud mask or precipitation). Although clear-sky radiance assimilation in LAMs still can be improved, many studies directly aim at assimilating cloud-affected radiances to maximise the use of available observations. Nevertheless, even without considering cloud-affected radiances, clear-sky radiance assimilation has the potential to improve short-term regional-scale forecasts of cloud-related parameters. Compared to variational DA methods, ensemble-based DA has the advantage of providing flow-dependent information about the model background error that may be especially useful for all-sky radiance assimilation. This matter has not yet been exploited using hybrid DA methods in LAMs.

Cloud property assimilation methods also improve short-term cloud forecasts, while a quantitative comparison of their outcome is rather difficult due to the diversity of methods and retrievals and their evaluation in individual case studies. The complexity extends from rather simple methods that primarily modify model cloud-top properties like cloud-top temperature via methods that also try to take the dynamical fields into consideration up to computationally expensive variational/ensemble-based methods. The latter are the minority since a common objective of cloud property assimilation methods is the possibility to create as complete cloud analyses as possible without large computational efforts. Existing cloud property assimilation methods are often targeted for specific LAMs, geographical regions or synoptic conditions and often depend on additional observations than those provided by geostationary satellites. Little efforts have been made in order to design and evaluate cloud property assimilation methods for LAMs that are applicable anywhere in the world and that exploit geostationary retrievals without the necessity of involving further observation types. Moreover, vertical columns are considered in many methods rather than the three-dimensional evolution of cloud properties and the dynamical implications of modified cloud analyses are often neglected.

Many DA systems for LAMs, especially operational ones, are tailored for a certain geographical region and the assimilation of the specific observations that are available for that region. Such systems are not readily applicable to all locations or satellites. In order to evaluate the full potential of developed data assimilation methods using geostationary satellite observations and limited-area models, future developments should be tested with different limited-area models and satellites. With the target of providing more accurate cloudiness forecasts for demanding applications like solar power forecasting that require frequently updated short-term forecasts, cloud analysis methods should be evaluated on a long-term basis, for different geographical regions (e.g. mid-latitudes and the tropics), and for locations where other observations are sparse in order to make the best use of geostationary satellites.

Several publications that are referred to in this paper tackle diverse issues of cloud DA by further developing the assimilation strategies and methods. Nevertheless, only some of the publications evaluate the methods at convection-permitting scales. The potential advantage of limited-area models of providing high-resolution forecasts of clouds in combination with geostationary satellite DA is thus unexploited so far.

The ability of a data assimilation system to assimilate cloudy observations is closely tied to the ability of the LAM to produce realistic clouds in the model background. Improvements of model parameterisation schemes, e.g. by introducing stochastic elements, are thus expected to positively influence DA performance and keep increments small. Likewise, improved radiative transfer models directly impact satellite DA performance and facilitates the assimilation of observations from new satellites.

New generations of geostationary satellites like Himawari-8, MTG and GOES-R bring more channels and higher temporal and spatial resolution. This imposes new challenges and brings more opportunities for data assimilation methods and NWP. It should lead to an increased exchange between researcher communities and institutions in order to use the newly available observations as efficiently as possible with the diverse LAMs that exist.

This review shows that both the assimilation of radiances and cloud properties in regional NWP offer great potential for short-term forecasting of cloudiness. The reason is that the topic comes along with diverse problems and challenges to be overcome, e.g. the choice of satellite channels or retrieved cloud properties and their optimal processing, the non-linearity of cloud processes, observation quality control, bias correction and subsampling, cloud classification and localisation, observation and background error estimation, computational efficiency, domain size, domain location and grid spacing optimisation.

The future lies in combined approaches that make the best possible use of available observations, radiance and cloud property assimilation methods and hybrid data assimilation techniques. The optimal use of cloud-free and cloud-affected radiance assimilation as well as available derived cloud properties will lead to more accurate cloud analyses and short-term cloudiness forecasts.

References

- ALBERS, S.C., J.A. MCGINLEY, D.L. BIRKENHEUER, J.R. SMART, 1996: The Local Analysis and Prediction System (LAPS): Analyses of Clouds, Precipitation, and Temperature. – Wea. Forecast. 11, 273–287, DOI:10.1175/ 1520-0434(1996)011<0273:TLAAPS>2.0.CO;2.
- ANDERSON, J., T. HOAR, K. RAEDER, H. LIU, N. COLLINS, R. TORN, A. AVELLANO, 2009: The Data Assimilation Research Testbed: A Community Facility. – Bull. Amer. Meteor. Soc. 90, 1283–1296, DOI:10.1175/2009BAMS2618.1.
- AULIGNÉ, T., 2014a: Multivariate Minimum Residual Method for Cloud Retrieval. Part I: Theoretical Aspects and Simulated Observation Experiments. – Mon. Wea. Rev. 142, 4383–4398, DOI:10.1175/MWR-D-13-00172.1.
- AULIGNÉ, T., 2014b: Multivariate Minimum Residual Method for Cloud Retrieval. Part II: Real Observations Experiments. – Mon. Wea. Rev. 142, 4399–4415, DOI: 10.1175/MWR-D-13-00173.1.
- BANNISTER, R.N., 2017: A review of operational methods of variational and ensemble-variational data assimilation. – Quart. J. Roy. Meteor. Soc. 143, 607–633, DOI: 10.1002/qj.2982.
- BARKER, D., X.-Y. HUANG, Z. LIU, T. AULIGNÉ, X. ZHANG, S. RUGG, R. AJJAJI, A. BOURGEOIS, J. BRAY, Y. CHEN, M. DEMIRTAS, Y.-R. GUO, T. HENDERSON, W. HUANG, H.-C. LIN, J. MICHALAKES, S. RIZVI, X. ZHANG, 2012: The Weather Research and Forecasting Model's Community Variational/Ensemble Data Assimilation System: WRFDA. – Bull. Amer. Meteor. Soc. 93, 831–843, DOI:10.1175/ BAMS-D-11-00167.1.
- BAUER, P., T. AULIGNÉ, W. BELL, A. GEER, V. GUIDARD, S. HEILLIETTE, M. KAZUMORI, M.-J. KIM, E.H.-C. LIU, A.P. MCNALLY, B. MACPHERSON, K. OKAMOTO, R. RENSHAW, L.-P. RIISHØJGAARD, 2011a: Satellite cloud and precipitation assimilation at operational NWP centres. – Quart. J. Roy. Meteor. Soc. 137, 1934–1951, DOI:10.1002/qj.905.
- BAUER, P., G. OHRING, C. KUMMEROW, T. AULIGNE, 2011b: Assimilating Satellite Observations of Clouds and Precipitation into NWP Models. – Bull. Amer. Meteor. Soc. 92, ES25-ES28, DOI:10.1175/2011BAMS3182.1.
- BAYLER, G.M., R.M. AUNE, W.H. RAYMOND, 2000: NWP Cloud Initialization Using GOES Sounder Data and Improved Modeling of Nonprecipitating Clouds. – Mon. Wea. Rev. **128**, 3911–3920, DOI:10.1175/ 1520-0493(2001)129<3911:NCIUGS>2.0.CO;2.

- BENGTSSON, L., M. STEINHEIMER, P. BECHTOLD, J.-F. GELEYN, 2013: A stochastic parametrization for deep convection using cellular automata. – Quart. J. Roy. Meteor. **139**, 1533–1543, DOI:10.1002/qj.2108.
- BENJAMIN, S.G., D. DÉVÉNYI, S.S. WEYGANDT, K.J. BRUNDAGE, J.M. BROWN, G.A. GRELL, D. KIM, B.E. SCHWARTZ, T.G. SMIRNOVA, T.L. SMITH, G.S. MANIKIN, 2004: An Hourly Assimilation–Forecast Cycle: The RUC. – Mon. Wea. Rev. 132, 495–518, DOI:10.1175/ 1520-0493(2004)132<0495:AHACTR>2.0.CO;2.
- BENJAMIN, S., C. ALEXANDER, S. WEYGANDT, D. DOW-ELL, M. HU, T. ALCOTT, T. SMIRNOVA, J. OLSON, J. BROWN, E. JAMES, J. KENYON, 2016a: Improvements in HRRRv2/RAPv3 for Aug 2016 Implementation at NCEP for More Accurate Warm-Season and Cold-Season NWP. – NOAA Earth System Research Laboratory, Boulder, CO. https://rapidrefresh.noaa.gov/: accessed 15 January 2017.
- BENJAMIN, S.G., S.S. WEYGANDT, J.M. BROWN, M. HU, C.R. ALEXANDER, T.G. SMIRNOVA, J.B. OLSON, E.P. JAMES, D.C. DOWELL, G.A. GRELL, H. LIN, S.E. PECKHAM, T.L. SMITH, W.R. MONINGER, J.S. KENYON, G.S. MANIKIN, 2016b: A North American Hourly Assimilation and Model Forecast Cycle: The Rapid Refresh. – Mon. Wea. Rev. 144, 1669–1694, DOI:10.1175/MWR-D-15-0242.1.
- BENNARTZ, R., T. GREENWALD, 2011: Current problems in scattering radiative transfer modelling for data assimilation. – Quart. J. Roy. Meteor. 137, 1952–1962, DOI:10.1002/qj.953.
- BROUSSEAU, P., Y. SEITY, D. RICARD, J. LÉGER, 2016: Improvement of the forecast of convective activity from the AROME-France system. – Quart. J. Roy. Meteor. 142, 2231–2243, DOI: 10.1002/qj.2822.
- CHEN, Y., H. WANG, J. MIN, X.-Y. HUANG, P. MINNIS, R. ZHANG, J. HAGGERTY, R. PALIKONDA, 2015: Variational Assimilation of Cloud Liquid/Ice Water Path and Its Impact on NWP. – J. Appl. Meteor. Climatol. 54, 1809–1825, DOI: 10.1175/JAMC-D-14-0243.1.
- CHEN, Y., R. ZHANG, D. MENG, J. MIN, L. ZHANG, 2016: Variational assimilation of satellite cloud water/ice path and microphysics scheme sensitivity to the assimilation of a rainfall case. – Adv. Atmos. Sci. 33, 1158–1170, DOI: 10.1007/s00376-016-6004-3.
- CINTINEO, R., J.A. OTKIN, M. XUE, F. KONG, 2014: Evaluating the Performance of Planetary Boundary Layer and Cloud Microphysical Parameterization Schemes in Convection-Permitting Ensemble Forecasts Using Synthetic GOES-13 Satellite Observations. – Mon. Wea. Rev. 142, 163–182, DOI: 10.1175/MWR-D-13-00143.1.
- CINTINEO, R.M., J.A. OTKIN, T.A. JONES, S. KOCH, D.J. STENS-RUD, 2016: Assimilation of Synthetic GOES-R ABI Infrared Brightness Temperatures and WSR-88D Radar Observations in a High-Resolution OSSE. – Mon. Wea. Rev. **144**, 3159–3180, DOI:10.1175/MWR-D-15-0366.1.
- CROS, S., N. SÉBASTIEN, O. LIANDRAT, N. SCHMUTZ, 2014: Cloud pattern prediction from geostationary meteorological satellite images for solar energy forecasting. – In: Proc. SPIE. 9242, Remote Sensing of Clouds and the Atmosphere XIX; and Optics in Atmospheric Propagation and Adaptive Systems XVII, 924202. DOI:10.1117/12.2066853.
- CROS, S., M. TURPIN, S. MADEC, S. POULENARD, M. CROSNIER, B. ROY, C. BERTIN, 2015: Day time cloud forecasting for space-to-ground optical communication optimization. – 2015 EUMETSAT Meteorological Satellite Conference, Toulouse, France.
- DE HAAN, S., S.H. VAN DER VEEN, 2014: Cloud Initialization in the Rapid Update Cycle of HIRLAM. – Wea. Forecast. 29, 1120–1133, DOI:10.1175/WAF-D-13-00071.1.

- DERRIEN, M., H. LE GLÉAU, 2005: MSG/SEVIRI cloud mask and type from SAFNWC. – Int. J. Remote Sens. 26, 4707–4732, DOI:10.1080/01431160500166128.
- DESCOMBES, G., T. AULIGNÉ, H.-C. LIN, D. XU, C. SCHWARTZ, F. VANDENBERGHE, 2014: Multi-sensor Advection Diffusion nowCast (MADCast) for cloud analysis and short-term prediction. – NCAR Technical Note NCAR/TN-509+STR. UCAR/NCAR.
- DIAGNE, M., M. DAVID, P. LAURET, J. BOLAND, N. SCHMUTZ, 2013: Review of solar irradiance forecasting methods and a proposition for small-scale insular grids. – Renew. Sustainable Energy Rev. 27, 65–76, DOI:10.1016/j.rser.2013.06.042.
- ERRICO, R.M., L. FILLION, D. NYCHKA, Z.-Q. LU, 2000: Some statistical considerations associated with the data assimilation of precipitation observations. – Quart. J. Roy. Meteor. **126**, 339–359, DOI:10.1002/qj.49712656217.
- ERRICO, R.M., P. BAUER, J.-F. MAHFOUF, 2007a: Issues Regarding the Assimilation of Cloud and Precipitation Data. – J. Atmos. Sci. 64, 3785–3798, DOI:10.1175/2006JAS2044.1.
- ERRICO, R.M., G. OHRING, F. WENG, P. BAUER, B. FER-RIER, J.-F. MAHFOUF, J. TURK, 2007b: Assimilation of Satellite Cloud and Precipitation Observations in Numerical Weather Prediction Models: Introduction to the JAS Special Collection. – J. Atmos. Sci. 64, 3737–3741, DOI: 10.1175/2007JAS2622.1.
- FORMAN, B.A., S.A. MARGULIS, 2010: Assimilation of multiresolution radiation products into a downwelling surface radiation model: 2. Posterior ensemble implementation. – J. Geophys. Res. 115, D22116, DOI:10.1029/2010JD013950.
- GEER, A.J., P. BAUER, 2011: Observation errors in all-sky data assimilation. – Quart. J. Roy. Meteor. 137, 2024–2037, DOI: 10.1002/qj.830.
- GEER, A.J., K. LONITZ, P. WESTON, M. KAZUMORI, K. OKA-MOTO, Y. ZHU, E.H. LIU, A. COLLARD, W. BELL, S. MI-GLIORINI, P. CHAMBON, N. FOURRIÉ, M.-J. KIM, C. KÖPKEN-WATTS, C. SCHRAFF, 2017: All-sky satellite data assimilation at operational weather forecasting centres. – Quart. J. Roy. Meteor., published online November 14th, 2017; DOI: 10.1002/qj.3202.
- GREENWALD, T.J., R. HERTENSTEIN, T. VUKIĆEVIĆ, 2002: An All-Weather Observational Operator for Radiance Data Assimilation with Mesoscale Forecast Models. – Mon. Wea. Rev. 130, 1882–1897, DOI:10.1175/ 1520-0493(2002)130<1882:AAWOOF>2.0.CO;2.
- GREENWALD, T.J., T. VUKIĆEVIĆ, L.D. GRASSO, T.H. VONDER HAAR, 2004: Adjoint sensitivity analysis of an observational operator for visible and infrared cloudy-sky radiance assimilation. – Quart. J. Roy. Meteor. **130**, 685–705, DOI: 10.1256/qj.03.44.
- GUIDARD, V., N. FOURRIÉ, 2010: Status of cloudy infrared radiance assimilation at Météo-France. – In: Proceedings of the ECMWF-JCSDA Workshop on Assimilating Satellite Observations of Clouds and Precipitation into NWP Models, European Centre for Medium-Range Weather Forecasts, Reading, UK.
- GUIDARD, V., C. FISCHER, M. NURET, A. DŽIEDŽIC, 2006: Evaluation of the ALADIN 3D-VAR with observations of the MAP campaign. – Meteor. Atmos. Phys. 92, 161–173, DOI: 10.1007/s00703-005-0156-5.
- GUSTAFSSON, N., L. BERRE, S. HORNQUIST, X.-Y. HUANG, M. LINDSKOG, B. NAVASCUES, K.S. MOGENSEN, S. THOR-STEINSSON, 2001: Three-dimensional variational data assimilation for a limited area model. Part I: General formulation and the background error constraint. – Tellus A 53, 425–446, DOI:10.1111/j.1600-0870.2001.00425.x.
- GUSTAFSSON, N., T. JANJIĆ, C. SCHRAFF, D. LEUENBERGER, M. WEISSMANN, H. REICH, P. BROUSSEAU, T. MONTMERLE,

E. WATTRELOT, A. BUČÁNEK, M. MILE, R. HAMDI, M. LIND-SKOG, J. BARKMEIJER, M. DAHLBOM, B. MACPHERSON, S. BALLARD, G. INVERARITY, J. CARLEY, C. ALEXANDER, D. DOWELL, S. LIU, Y. IKUTA, T. FUJITA, 2018: Survey of data assimilation methods for convective-scale numerical weather prediction at operational centres. – Quart. J. Roy. Meteor., published online March 30th, 2018; DOI:10.1002/qj.3179.

- HAIDEN, T., J. TRENTMANN, 2015: Verification of cloudiness and radiation forecasts in the greater Alpine region. – Meteorol. Z. 25, 3–15, DOI:10.1127/metz/2015/0630.
- HARNISCH, F., M. WEISSMANN, Á. PERIÁÑEZ, 2016: Error model for the assimilation of cloud-affected infrared satellite observations in an ensemble data assimilation system. – Quart. J. Roy. Meteor. **142**, 1797–1808, DOI:10.1002/qj.2776.
- HOHENEGGER, C., C. SCHÄR, 2007: Predictability and Error Growth Dynamics in Cloud-Resolving Models. – J. Atmos. Sci. 64, 4467–4478, DOI:10.1175/2007JAS2143.1.
- HOUTEKAMER, P.L., F. ZHANG, 2016: Review of the Ensemble Kalman Filter for Atmospheric Data Assimilation. – Mon. Wea. Rev. 144, 4489–4532, DOI:10.1175/ MWR-D-15-0440.1.
- HU, M., S. WEYG, M. XUE, S. BENJAMIN, 2007: Development and Testing of a New Cloud Analysis Package using Radar, Satellite, and Surface Cloud Observations within GSI for Initializing Rapid Refresh. In: Proceedings of the 18th Conference on Numerical Weather Prediction. – American Meteorological Society, Park City, UT, USA.
- INMAN, R.H., H.T.C. PEDRO, C.F.M. COIMBRA, 2013: Solar forecasting methods for renewable energy integration. – Prog. Energ. Combust. 39, 535–576, DOI:10.1016/j.pecs.2013.06.002.
- JIMENEZ, P.A., J.P. HACKER, J. DUDHIA, S. ELLEN HAUPT, J.A. RUIZ-ARIAS, C.A. GUEYMARD, G. THOMPSON, T. EID-HAMMER, A. DENG, 2016: WRF-Solar: Description and Clear-Sky Assessment of an Augmented NWP Model for Solar Power Prediction. – Bull. Amer. Meteor. Soc. 97, 1249–1264, DOI:10.1175/BAMS-D-14-00279.1.
- JONES, T.A., J.A. OTKIN, D.J. STENSRUD, K. KNOPFMEIER, 2013a: Assimilation of Satellite Infrared Radiances and Doppler Radar Observations during a Cool Season Observing System Simulation Experiment. – Mon. Wea. Rev. 141, 3273–3299, DOI:10.1175/MWR-D-12-00267.1.
- JONES, T.A., D.J. STENSRUD, P. MINNIS, R. PALIKONDA, 2013b: Evaluation of a Forward Operator to Assimilate Cloud Water Path into WRF-DART. – Mon. Wea. Rev. **141**, 2272–2289, DOI:10.1175/MWR-D-12-00238.1.
- JONES, T.A., J.A. OTKIN, D.J. STENSRUD, K. KNOPFMEIER, 2014: Forecast Evaluation of an Observing System Simulation Experiment Assimilating Both Radar and Satellite Data. – Mon. Wea. Rev. 142, 107–124, DOI:10.1175/MWR-D-13-00151.1.
- JONES, T.A., D. STENSRUD, L. WICKER, P. MINNIS, R. PA-LIKONDA, 2015: Simultaneous Radar and Satellite Data Storm-Scale Assimilation Using an Ensemble Kalman Filter Approach for 24 May 2011. – Mon. Wea. Rev. 143, 165–194, DOI:10.1175/MWR-D-14-00180.1.
- JONES, T.A., S. KOCH, Z. LI, 2017: Assimilating synthetic hyperspectral sounder temperature and humidity retrievals to improve severe weather forecasts. – Atmos. Res. 186, 9–25, DOI: 10.1016/j.atmosres.2016.11.004.
- KALNAY, E., 2003: Atmospheric Modeling, Data Assimilation, and Predictability. – Cambridge University Press, New York.
- KALNAY, E., H. LI, T. MIYOSHI, S.-C. YANG, J. BALLABRERA-POY, 2007: 4-D-Var or ensemble Kalman filter? – Tellus A 59, 758–773, DOI:10.1111/j.1600-0870.2007.00261.x.
- KOSTKA, P.M., M. WEISSMANN, R. BURAS, B. MAYER, O. STILLER, 2014: Observation Operator for Visible and Near-Infrared Satellite Reflectances. – J. Atmos. Oceanic Technol. 31, 1216–1233, DOI:10.1175/JTECH-D-13-00116.1.

- KUMAR, P., A.K. VARMA, 2016: Assimilation of INSAT-3D Hydro-Estimator Method Retrieved Rainfall for Short Range Weather Prediction. – Quart. J. Roy. Meteor. 143, 384–394, DOI:10.1002/qj.2929.
- LAUWAET, D., K. DE RIDDER, P. PANDEY, 2011: Assimilating remotely sensed cloud optical thickness into a mesoscale model. – Atmos. Chem. Phys. 11, 10269–10281, DOI: 10.5194/acp-11-10269-2011.
- LINDSKOG, M., N. GUSTAFSSON, B. NAVASCUES, K.S. MO-GENSEN, X.-Y. HUANG, X. YANG, U. ANDRAE, L. BERRE, S. THORSTEINSSON, J. RANTAKOKKO, 2001: Threedimensional variational data assimilation for a limited area model. Part II: Observation handling and assimilation experiments. – Tellus A 53, 447–468, DOI:10.1111/ j.1600-0870.2001.00447.x.
- LIPTON, A.E., 1993: Cloud Shading Retrieval and Assimilation in a Satellite-Model Coupled Mesoscale Analysis System. – Mon. Wea. Rev. **121**, 3062–3081, DOI:10.1175/ 1520-0493(1993)121<3062:CSRAAI>2.0.CO;2.
- LIPTON, A.E., G.D. MODICA, 1999: Assimilation of Visible-Band Satellite Data for Mesoscale Forecasting in Cloudy Conditions. – Mon. Wea. Rev. **127**, 265–278, DOI:10.1175/ 1520-0493(1999)127<0265:AOVBSD>2.0.CO;2.
- LIPTON, A.E., T.H. VONDER HAAR, 1990a: Preconvective Mesoscale Analysis over Irregular Terrain with a Satellite-Model Coupled System. – Mon. Wea. Rev. **118**, 1330–1358, DOI: 10.1175/1520-0493(1990)118<1330:PMAOIT>2.0.CO;2.
- LIPTON, A.E., T.H. VONDER HAAR, 1990b: Mesoscale Analysis by Numerical Modeling Coupled with Sounding Retrieval from Satellites. – Mon. Wea. Rev. **118**, 1308–1329, DOI: 10.1175/1520-0493(1990)118<1308:MABNMC>2.0.CO;2.
- LÓPEZ-COTO, I., J.L. BOSCH, P. MATHIESEN, J. KLEISSL, 2013: Comparison between several parameterization schemes in WRF for solar forecasting in coastal zones. – In: SOLAR Conference Proceedings. Baltimore, Maryland, USA.
- LORENZ, E., J. KÜHNERT, D. HEINEMANN, 2014: Overview of Irradiance and Photovoltaic Power Prediction, pp. 429–454. – In: TROCCOLI, A., L. DUBUS, S.E. HAUPT (Eds), Weather Matters for Energy. – Springer New York.
- MACPHERSON, B., B.J. WRIGHT, W.H. HAND, A.J. MAY-COCK, 1996: The Impact of MOPS Moisture Data in the U.K. Meteorological Office Mesoscale Data Assimilation Scheme. – Mon. Wea. Rev. **124**, 1746–1766, DOI:10.1175/ 1520-0493(1996)124<1746:TIOMMD>2.0.CO;2.
- MATHIESEN, P., C. COLLIER, J. KLEISSL, 2013: A high-resolution, cloud-assimilating numerical weather prediction model for solar irradiance forecasting. – Solar Energy 92, 47–61, DOI: 10.1016/j.solener.2013.02.018.
- MCNALLY, A.P., 2002: A note on the occurrence of cloud in meteorologically sensitive areas and the implications for advanced infrared sounders. – Quart. J. Roy. Meteor. 128, 2551–2556, DOI:10.1256/qj.01.206.
- MCNIDER, R.T., J.A. SONG, S.Q. KIDDER, 1995: Assimilation of GOES-derived solar insolation into a mesoscale model for studies of cloud shading effects. – Int. J. Remote Sens. 16, 2207–2231, DOI:10.1080/01431169508954552.
- MEETSCHEN, D., B.J.J.M. VAN DEN HURK, F. AMENT, M. DR-USCH, 2004: Optimized Surface Radiation Fields Derived from Meteosat Imagery and a Regional Atmospheric Model. – J. Hydrometeorol. 5, 1091–1101, DOI:10.1175/JHM-389.1.
- MIGLIORINI, S., 2012: On the Equivalence between Radiance and Retrieval Assimilation. – Mon. Wea. Rev. **140**, 258–265, DOI: 10.1175/MWR-D-10-05047.1.
- MIGLIORINI, S., C. PICCOLO, C.D. RODGERS, 2008: Use of the Information Content in Satellite Measurements for an Efficient Interface to Data Assimilation. – Mon. Wea. Rev. 136, 2633–2650, DOI:10.1175/2007MWR2236.1.

- MINNIS, P., L. NGUYEN, R. PALIKONDA, P.W. HECK, D.A. SPAN-GENBERG, D.R. DOELLING, J.K. AYERS, J. SMITH WILLIAM L., M.M. KHAIYER, Q.Z. TREPTE, L.A. AVEY, F.-L. CHANG, C.R. YOST, T.L. CHEE, S.-M. SZEDUNG, 2008: Near-real time cloud retrievals from operational and research meteorological satellites. – In: Proc. SPIE 7107. Remote Sensing of Clouds and the Atmosphere XIII, 710703. DOI:10.1117/12.800344.
- NOAA NCEI, 2017: Rapid Refresh (RAP). National Centers for Environmental Information (NCEI) formerly known as National Climatic Data Center (NCDC). – https://www. ncdc.noaa.gov/data-access/model-data/model-datasets/ rapid-refresh-rap (accessed 8 January 2017).
- OKAMOTO, K., 2017: Evaluation of IR radiance simulation for all-sky assimilation of Himawari-8/AHI in a mesoscale NWP system: Evaluation of All-Sky IR Radiance of Himawari-8. – Quart. J. Roy. Meteor. 143, 1517–1527, DOI:10.1002/qj.3022.
- OTKIN, J.A., 2010: Clear and cloudy sky infrared brightness temperature assimilation using an ensemble Kalman filter. – J. Geophys. Res. 115, D19207, DOI:10.1029/2009JD013759.
- OTKIN, J.A., 2012a: Assimilation of water vapor sensitive infrared brightness temperature observations during a high impact weather event. – J. Geophys. Res. 117, D19203, DOI: 10.1029/2012JD017568.
- OTKIN, J.A., 2012b: Assessing the impact of the covariance localization radius when assimilating infrared brightness temperature observations using an ensemble Kalman filter. – Mon. Wea. Rev. **140**, 543–561, DOI:10.1175/MWR-D-11-00084.1.
- OTKIN, J.A., T.J. GREENWALD, 2008: Comparison of WRF Model-Simulated and MODIS-Derived Cloud Data. – Mon. Wea. Rev. 136, 1957–1970, DOI:10.1175/2007MWR2293.1.
- OTKIN, J.A., W.E. LEWIS, A.J. LENZEN, B.D. MCNOLDY, S.J. MAJUMDAR, 2017: Assessing the Accuracy of the Cloud and Water Vapor Fields in the Hurricane WRF (HWRF) Model Using Satellite Infrared Brightness Temperatures. – Mon. Wea. Rev. 145, 2027–2046, DOI:10.1175/ MWR-D-16-0354.1.
- OTKIN, J.A., R. POTTHAST, A. LAWLESS, 2018: Nonlinear bias correction for satellite data assimilation using Taylor series polynomials. – Mon. Wea. Rev., published online. DOI: 10.1175/MWR-D-17-0171.1.
- PINCUS, R., R.J. PATRICK HOFMANN, J.L. ANDERSON, K. RAE-DER, N. COLLINS, J.S. WHITAKER, 2011: Can Fully Accounting for Clouds in Data Assimilation Improve Short-Term Forecasts by Global Models? – Mon. Wea. Rev. **139**, 946–957, DOI:10.1175/2010MWR3412.1.
- POLKINGHORNE, R., T. VUKICEVIC, 2011: Data Assimilation of Cloud-Affected Radiances in a Cloud-Resolving Model. – Mon. Wea. Rev. **139**, 755–773, DOI:10.1175/ 2010MWR3360.1.
- POLKINGHORNE, R., T. VUKICEVIC, K.F. EVANS, 2010: Validation of Cloud-Resolving Model Background Data for Cloud Data Assimilation. – Mon. Wea. Rev. 138, 781–795, DOI:10.1175/ 2009MWR3012.1.
- QIN, Z., X. ZOU, F. WENG, 2013: Evaluating Added Benefits of Assimilating GOES Imager Radiance Data in GSI for Coastal QPFs. – Mon. Wea. Rev. 141, 75–92, DOI:10.1175/ MWR-D-12-00079.1.
- RAYMOND, W.H., G.S. WADE, T.H. ZAPOTOCNY, 2004: Assimilating GOES Brightness Temperatures. Part I: Upper-Tropospheric Moisture. – J. Appl. Meteor. 43, 17–27, DOI: 10.1175/1520-0450(2004)043<0017:AGBTPI>2.0.CO;2.
- RENSHAW, R., P.N. FRANCIS, 2011: Variational assimilation of cloud fraction in the operational Met Office Unified Model. – Quart. J. Roy. Meteor. 137, 1963–1974, DOI:10.1002/qj.980.
- RUGGIERO, F.H., K.D. SASHEGYI, A.E. LIPTON, R.V. MADALA, S. RAMAN, 1999: Coupled Assimilation of Geostationary Satellite Sounder Data into a

Mesoscale Model Using the Bratseth Analysis Approach. – Mon. Wea. Rev. **127**, 802–821, DOI:10.1175/1520-0493(1999)127<0802:CAOGSS>2.0.CO;2.

- SCHIPPER, J., P. MATHIESEN, 2015: Advanced Cloud Simulations for Improved Dayahead Solar Power Forecasts. – 5th Solar Integration Workshop, Brussels, Belgium.
- SCHOMBURG, A., C. SCHRAFF, R. POTTHAST, 2014: A concept for the assimilation of satellite cloud information in an Ensemble Kalman Filter: single-observation experiments. – Quart. J. Roy. Meteor. 141, 893–908, DOI:10.1002/qj.2407.
- SCHRAFF, C., H. REICH, A. RHODIN, A. SCHOMBURG, K. STEPHAN, A. PERIÁÑEZ, R. POTTHAST, 2016: Kilometrescale ensemble data assimilation for the COSMO model (KENDA). – Quart. J. Roy. Meteor. 142, 1453–1472, DOI: 10.1002/qj.2748.
- SEAMAN, C.J., M. SENGUPTA, T.H. VONDER HAAR, 2010: Mesoscale satellite data assimilation: impact of cloud-affected infrared observations on a cloud-free initial model state. – Tellus A 62, 298–318, DOI:10.1111/j.1600-0870.2010.00436.x.
- SENGUPTA, M., A. HABTE, S. KURTZ, A. DOBOS, S. WILBERT, E. LORENZ, T. STOFFEL, D. RENNÉ, C. GUEYMARD, D. MY-ERS, S. WILCOX, P. BLANC, R. PEREZ, 2015: Best Practices Handbook for the Collection and Use of Solar Resource Data for Solar Energy Applications. – National Renewable Energy Laboratory (NREL).
- SHAO, H., J. DERBER, X.-Y. HUANG, M. HU, K. NEW-MAN, D. STARK, M. LUEKEN, C. ZHOU, L. NANCE, Y.-H. KUO, B. BROWN, 2016: Bridging Research to Operations Transitions: Status and Plans of Community GSI. – Bull. Amer. Meteor. Soc. 97, 1427–1440, DOI:10.1175/ BAMS-D-13-00245.1.
- SINGH, R., P.K. PAL, P.C. JOSHI, 2010: Assimilation of Kalpana very high resolution radiometer water vapor channel radiances into a mesoscale model. – J. Geophys. Res. **115**, D18124, DOI:10.1029/2010JD014027.
- SINGH, R., S.P. OJHA, C.M. KISHTAWAL, P.K. PAL, A.S. KIRAN KUMAR, 2016: Impact of the assimilation of INSAT-3D radiances on short-range weather forecasts. – Quart. J. Roy. Meteor. **142**, 120–131, DOI:10.1002/qj.2636.
- SOMMER, M., M. WEISSMANN, 2014: Observation impact in a convective-scale localized ensemble transform Kalman filter: Observation Impact in a Convective-Scale LETKF. – Quart. J. Roy. Meteor. 140, 2672–2679, DOI:10.1002/qj.2343.
- SOUTO, M.J., C.F. BALSEIRO, V. PÉREZ-MUÑUZURI, M. XUE, K. BREWSTER, 2003: Impact of Cloud Analysis on Numerical Weather Prediction in the Galician Region of Spain. – J. Appl. Meteor. 42, 129–140, DOI:10.1175/ 1520-0450(2003)042<0129:IOCAON>2.0.CO;2.
- STENGEL, M., P. UNDÉN, M. LINDSKOG, P. DAHLGREN, N. GUSTAFSSON, R. BENNARTZ, 2009: Assimilation of SEVIRI infrared radiances with HIRLAM 4D-Var. – Quart. J. Roy. Meteor. 135, 2100–2109, DOI:10.1002/qj.501.
- STENGEL, M., M. LINDSKOG, P. UNDÉN, N. GUSTAFSSON, R. BENNARTZ, 2010: An extended observation operator in HIRLAM 4D-VAR for the assimilation of cloud-affected satellite radiances. – Quart. J. Roy. Meteor. 136, 1064–1074, DOI:10.1002/qj.621.
- STENGEL, M., M. LINDSKOG, P. UNDÉN, N. GUSTAFSSON, 2013: The impact of cloud-affected IR radiances on forecast accuracy of a limited-area NWP model. – Quart. J. Roy. Meteor. 139, 2081–2096, DOI:10.1002/qj.2102.
- TAYLOR, R.B.E., R.J. RENSHAW, R.W. SAUNDERS, P.N. FRANCIS, 2008: Assimilation of SEVIRI cloud-top parameters in the Met Office regional forecast model. – Met Office, Exeter, UK.
- VAN DER VEEN, S.H., 2013: Improving NWP Model Cloud Forecasts Using Meteosat Second-Generation Im-

agery. – Mon. Wea. Rev. **141**, 1545–1557, DOI:10.1175/ MWR-D-12-00021.1.

- VAN LEEUWEN, P.J., 2001: An Ensemble Smoother with Error Estimates. – Mon. Wea. Rev. **129**, 709–728, DOI:10.1175/ 1520-0493(2001)129<0709:AESWEE>2.0.CO;2.
- VUKICEVIC, T., T. GREENWALD, M. ZUPANSKI, D. ZUPANSKI, T. VONDER HAAR, A.S. JONES, 2004: Mesoscale Cloud State Estimation from Visible and Infrared Satellite Radiances. – Mon. Wea. Rev. **132**, 3066–3077, DOI:10.1175/MWR2837.1.
- VUKICEVIC, T., M. SENGUPTA, A.S. JONES, T. VONDER HAAR, 2006: Cloud-Resolving Satellite Data Assimilation: Information Content of IR Window Observations and Uncertainties in Estimation. – J. Atmos. Sci. 63, 901–919, DOI: 10.1175/JAS3639.1.
- WATTS, P.D., R. BENNARTZ, F. FELL, 2011: Retrieval of twolayer cloud properties from multispectral observations using optimal estimation. – J. Geophys. Res. **116**, D16203, DOI: 10.1029/2011JD015883.
- WENG, F., 2007: Advances in Radiative Transfer Modeling in Support of Satellite Data Assimilation. – J. Atmos. Sci. 64, 3799–3807, DOI:10.1175/2007JAS2112.1.
- WEYGANDT, S., S.G. BENJAMIN, D. DÉVÉNYI, J.M. BROWN, P. MINNIS, 2006: Cloud and hydrometeor analysis using metar, radar, and satellite data within the RUC/Rapid-Refresh model. – In: Proceedings of the 12th Conference on Aviation Range and Aerospace Meteorology. – American Metrological Society, Atlanta, GA, USA.
- WHITE, A.T., A. POUR-BIAZAR, K. DOTY, B. DORNBLASER, R.T. MCNIDER, 2018: Improving Cloud Simulation for Air Quality Studies Through Assimilation of Geostationary Satellite Observations in Retrospective Meteorological Modeling. – Mon. Wea. Rev. 146, 29–48, DOI:10.1175/ MWR-D-17-0139.1.
- XU, D., T. AULIGNÉ, X.-Y. HUANG, 2015: A validation of the multivariate and minimum residual method for cloud retrieval using radiance from multiple satellites. – Adv. Atmos. Sci. 32, 349–362, DOI:10.1007/s00376-014-3258-5.
- YANG, H., J. KLEISSL, 2016: Preprocessing WRF initial conditions for coastal stratocumulus forecasting. – Solar Energy 133, 180–193, DOI:10.1016/j.solener.2016.04.003.
- YANG, C., Z. LIU, F. GAO, P.P. CHILDS, J. MIN, 2017: Impact of assimilating GOES imager clear-sky radiance with a rapid refresh assimilation system for convection-permitting forecast over Mexico. – J. Geophys. Res. Atmos. **122**, 5472–5490, DOI:10.1002/2016JD026436.
- YUCEL, I., W.J. SHUTTLEWORTH, R.T. PINKER, L. LU, S. SOROOSHIAN, 2002: Impact of Ingesting Satellite-Derived Cloud Cover into the Regional Atmospheric Modeling System. – Mon. Wea. Rev. **130**, 610–628, DOI:10.1175/ 1520-0493(2002)130<0610:IOISDC>2.0.CO;2.
- YUCEL, I., W. JAMES SHUTTLEWORTH, X. GAO, S. SOROOSHIAN, 2003: Short-Term Performance of MM5 with Cloud-Cover Assimilation from Satellite Observations. – Mon. Wea. Rev. 131, 1797–1810, DOI:10.1175//2565.1.
- ZAPOTOCNY, T.H., W.P. MENZEL, J.A. JUNG, J.P. NELSON, 2005: A Four-Season Impact Study of Rawinsonde, GOES, and POES Data in the Eta Data Assimilation System. Part I: The Total Contribution. – Wea. Forecast. **20**, 161–177, DOI: 10.1175/WAF837.1.
- ZHU, M., V. LEEUWEN, P. J, W. ZHANG, 2017: Estimating model error covariances using particle filters. – Quart. J. Roy. Meteor., published online 8th November, 2017; DOI: 10.1002/qj.3132.
- ZOU, X., C. DA, 2014: An objective regional cloud mask algorithm for GOES infrared imager radiance assimilation. – J. Geophys. Res. Atmos. **119**, 6666–6680, DOI: 10.1002/2014JD021455.

- ZOU, X., Z. QIN, F. WENG, 2011: Improved Coastal Precipitation Forecasts with Direct Assimilation of GOES-11/12 Imager Radiances. – Mon. Wea. Rev. 139, 3711–3729, DOI: 10.1175/MWR-D-10-05040.1.
- ZUPANSKI, D., M. ZUPANSKI, L.D. GRASSO, R. BRUMMER, I. JANKOV, D. LINDSEY, M. SENGUPTA, M. DEMARIA, 2011: Assimilating synthetic GOES-R radiances in cloudy conditions using an ensemble-based method. – Int. J. Remote Sens. 32, 9637–9659, DOI:10.1080/01431161.2011.572094.